Power to Bias? The Effect of Attorney Empowerment in Voir Dire on Jury Prejudice and Race^{*†}

Jee-Yeon K. Lehmann[‡] University of Houston Jeremy B. Smith[§] Analysis Group

October 29, 2013

Abstract

Giving attorneys more power in the voir dire (jury selection) process may allow them to 1) more easily dismiss jurors whom they wish to strike on *a priori* grounds; 2) acquire information that enables them to identify favorably-inclined jurors more precisely; or both. Attorneys who are more skilled can better leverage their increased power to retain the jurors they prefer. We show theoretically that, since defense attorneys tend to prefer non-white jurors *a priori*, the interaction of empowerment and defense attorney skill should produce juries with a greater proportion of non-whites if only the first mechanism is operative, but need not have this effect if the second is operative. We find empirically that skilled and empowered attorneys can indeed stack juries by retaining jurors predisposed to their side at a greater rate. However, we find that empowerment and skill have small and insignificant impacts on the racial composition of the seated jury. In the context of our model, this result implies that, for at least some of the trials in our dataset, attorneys leveraged empowerment in voir dire to learn more about potential jurors, rather than simply to strike more effectively by relying on racial stereotypes. Our findings provide strong evidence that extensive voir dire involving attorneys can lead to the seating of biased juries when opposing counsels are unequally skilled, yet the presence of this bias may not be detected in the observable characteristics of seated juries.

^{*}We are grateful to Kevin Lang, Claudia Olivetti, Michael Manove, Aimee Chin, Chinhui Juhn, Andy Zuppann, Elaine Liu, Steven Craig, Richard Boylan, Randy Ellis, and Johannes Schmieder for excellent advice and guidance. Special thanks to Daniele Paserman, Wesley Yin, John Harris, Marc Rysman, and David Rossman for their helpful comments. Thanks to all of our colleagues who have given generously of their time and advice. The views presented are those of the authors and do not reflect those of Analysis Group. Any errors are our own.

[†]Updated drafts of this paper can be downloaded from http://www.uh.edu/~jlehman2

[‡]Corresponding author. Assistant Professor, Department of Economics, University of Houston. E-mail: jlehmann@uh. edu 211B McElhinney Hall, Houston, TX 77204-5019.

[§]The author is an Associate at Analysis Group, Inc., 111 Huntington Avenue, Tenth Floor, Boston, MA 02199. Research for this paper was undertaken when he was a PhD student at Boston University.

"The modern raison d'être of [in-depth voir dire] is not to discover disqualifying prejudice, but quite the opposite: to allow lawyers to discover jurors they think are secretly biased in their clients' favor."

- Hon. Morris B. Hoffman¹

1 Introduction

Voir dire, more commonly known as jury selection, is the process by which a pool of potential jurors is questioned and pared down to the seated jury that will eventually hear a trial.² The legal rationale for conducting voir dire, rather than simply drawing jurors randomly from the population, is to ensure that juries can act impartially and competently in rendering their verdicts. However, there is a widespread notion that voir dire practices employed in many trials grant attorneys too much freedom to shape the jury, leading instead to the seating of biased juries and to the systematic exclusion of certain groups from jury service. Thus, jury selection is often perceived as harming defendants by leading to unfair trial outcomes, harming potential jurors by facilitating their discriminatory removal on the basis of race or other characteristics, and generally detracting from the trustworthiness of the legal system.³

In this paper, we offer a novel theoretical and empirical assessment of what attorneys can actually accomplish in voir dire given varying degrees of institutional constraints placed on them in the courtroom. In particular, we focus on two research questions that are motivated by the legal community's main concerns with the process of jury selection. First, when attorneys are granted more freedoms in jury selection, will the jurors who make it through the jury selection process be biased against defendants or the prosecution? Second, will the seated jurors be less likely to belong to certain identifiable groups?

We first answer these questions with a simple model of attorney behavior. In the model, attorneys on a given side are assumed to have the objective of retaining jurors who are favorably predisposed to their side. Such predispositions are assumed to be correlated with an observable juror characteristic, for example, race. When attorneys are "unempowered" – that is, when they have limited freedoms in voir dire – they must pursue their objective by striking potential jurors according to this characteristic. In courtrooms where attorneys are given more latitude, we model empowerment as having two possible effects. First, it can improve the ability of attorneys to strike whichever potential jurors they choose ("striking ability effect"). Second, it can allow attorneys to acquire more accurate information about

¹ "Unnatural Selection." New York Times. March 7, 2006.

²A detailed description of the voir dire process is provided in Lehmann and Smith (2013).

 $^{^{3}}$ King (1993) discusses these categories of harm from the perspective of the evolution of related Supreme Court decisions. Much of the popular and scholarly discussion on these matters focuses on exclusion from juries based on race, and even more specifically, on blacks excluded from juries in favor of whites. We take a broad perspective on the potential to exclude jurors on the basis of any observable characteristics. However, we also recognize the prevalence and importance of the focus on race and adopt this focus ourselves in most of our empirical analysis.

the predispositions of potential jurors than can be inferred from the observable characteristic alone ("information effect"). We allow the magnitudes of both of these effects to depend positively on the skill of the attorneys on each side.

The model predicts that, when attorneys are empowered, more skillful attorneys will have greater success in retaining jurors favorably inclined to their side. Therefore, empowerment can indeed lead to a seated jury that is more prejudiced against the defendant than a randomly-drawn jury would be – but only when the prosecuting attorney holds a skill advantage over the defense attorney. Likewise, empowerment can lead to a seated jury that is prejudiced in favor of the defendant when the defense attorney holds a skill advantage.

However, predictions concerning the effect of empowerment on the distribution of seated jurors by the observable characteristic depend on whether the striking ability effect, the information effect, or both is at work. For example, if empowerment only increases striking ability and prosecutors have an *a priori* preference for white jurors, they will leverage empowerment to strike non-white jurors at a higher rate. But defense attorneys will likewise leverage empowerment to strike white jurors at a higher rate. If the defense attorney holds a skill advantage, the overall effect of empowerment will be a greater proportion of non-whites on the seated jury than in the pool of potential jurors. However, when the information revelation effect of empowerment is also present, the defense may be able to identify some white jurors who are preferred to some non-white jurors (and vice versa for the prosecutor). In this case, the interaction of empowerment and defense attorney skill need not lead to a higher proportion of non-whites on seated juries, and can, in fact, be associated with white jurors being retained at a very high rate.

Next, we test our theoretical predictions empirically, using a rich dataset on all non-capital felony trials in four large and diverse counties over parts of a two-year period. The dataset includes information on the freedoms that were granted to attorneys during jury selection, on attorney skill, on observable characteristics of jurors, on the self-reported pre-deliberation leanings of jurors, and on numerous other characteristics of the trial and the parties involved.

In relation to existing work and conventional wisdom in the legal community, the results of our empirical analysis are surprising. Attorney empowerment in jury selection, other things equal, is associated with a concentration of jurors who report having greater favoritism *towards* the defense prior to deliberations. This effect is almost entirely accounted for by trials in which the defense attorney held a skill advantage. We interpret this result as confirming the basic validity of our model and its predictions. However, in our data, neither skill nor empowerment nor their interaction is associated with any substantial impact on the average composition of seated juries by observable characteristics, including most notably, race. In the context of our model, this result implies that, for at least some of the trials in our dataset, attorneys leveraged empowerment in voir dire to learn more about potential jurors, rather than just to strike more effectively while still relying on racial stereotypes as a crutch.

1.1 Related Literature and Our Contributions

Our work connects three branches of literature to which economists have contributed. The first set of studies comprises research in labor economics examining the role of information and learning on workers' hiring or wages when employers statistically discriminate based on an easily observable characteristic. Lessons from these studies inform the basic setup of our model and help us interpret the empirical results. The second and third sets of studies encompass studies of jury selection and trial outcomes. The former comprises theoretical studies on attorney behavior in jury selection, and the latter includes empirical studies on the effect of race and other jury characteristics on trial outcomes.

Among the first group of studies examining the role of information in statistical discrimination, we consider two studies in particular – Autor and Scarborough (2008) and Altonji and Pierret (2001) – to be most relevant for our work. Autor and Scarborough (2008) consider the impact of job testing on the hiring of minority workers in a large national retail firm. They find that while job testing led to more productive workers, it had no measurable impact on the rate of minority hiring. The authors interpret their findings as suggestive evidence that job testing raised the precision of the worker screening mechanism without introducing additional negative information about the minority applicants. Altonji and Pierret (2001) extend Farber and Gibbons (1996) by testing the impact of employer learning about the importance of easily observable versus difficult-to-observe correlates of productivity on workers' wages as they gain experience. Using the 1979 cohort of the National Longitudinal Survey of the Youth, they find evidence that employers statistically discriminate based on education early but the coefficient on education in the wage regression decreases with experience while the coefficient on the unobserved AFQT rises. Although we cannot directly apply these results from the labor market to jury selection, where the defense and the prosecuting attorneys have directly opposing goals in shaping the jury, taken together, these two studies and basic models of statistical discrimination offer some broad lessons for our theoretical model. They suggest that in situations where attorneys use easily observable correlates of prejudice in panelists to inform their selection decisions, attorneys who are able to learn more accurate information about potential jurors' true, unobserved biases will be able to better maximize their objective function by seating jurors who are favorable to their side with greater probability. However, depending on the informativeness of the additional signal relative to the information revealed by easily observable correlates of biases, the composition of the jury by these observable characteristics may or may not be measurably impacted.

In the law and economics literature, there is a common and natural framework underlying much of the previous theoretical work modeling attorney behavior in jury selection. The foundation is an assumption that potential jurors are drawn from a unidimensional distribution of predispositions. More specifically, other things equal, the defense attorney will want to strike the potential jurors with the strongest inclinations towards the prosecution, and the prosecuting attorney will likewise want to strike the potential jurors with the strongest inclinations towards the defense. Brams and Davis (1978) explore this framework from a game theoretic perspective, and characterize attorney optimal responses when facing various constraints related to jury size, sequence of decisions, and number of strikes allowed.⁴ Flanagan (2013) extends the basic model presented in Brams and Davis by providing a more complete characterizations of the equilibria. Neilson and Winter (2000) make use of the same underlying framework to gauge the impact of allowing attorneys a greater number of strikes on two types of injustice: false convictions and false acquittals.⁵ A common prediction that arises from this literature is that the probability of a conviction will often be too high relative to a case in which juries are chosen randomly unless the defense attorney is allowed a greater number of strikes than the prosecuting attorney.

There have been several attempts to establish empirical evidence that certain observable juror characteristics are associated with different criminal trial verdicts. Anwar et al. (2012b) find that juries formed from all-white jury pools convict black defendants at a substantially and statistically significantly higher rate than they convict white defendants, while Anwar et al. (2012a) find that older jurors are more likely to convict. Lee (2010) uses state variation in the timing of jury reforms, and finds that greater potential racial heterogeneity on juries is associated with lower conviction rates of minority defendants. Our results in Lehmann and Smith (2013) show that juries with higher average income and religiousness are less likely to acquit in general, and that juries with a greater proportion of blacks are less likely to convict when both defendant and victim are also black. On the other hand, Hannaford-Agor et al. (2002) find little relation between jury racial composition and the probability of a hung jury.⁶

This empirical literature establishes that attorneys who wish to maximize the probability of a case being decided for their side *should want* to remove jurors on the basis of race and perhaps other observable characteristics, at least in the absence of other useful information. However, it has not satisfactorily explored how attorneys actually behave and what they can accomplish in voir dire.⁷ The theoretical literature, on the other hand, has not incorporated empirical insights on juror race comprehensively and can be usefully extended to consider their implications on jury composition. We

 $^{{}^{4}}$ Roth, Kadane and Degroot (1977), Degroot and Kadane (1980) and Kadane, Stone and Wallstrom (1999) provide technical extensions on this theme.

⁵Kadane and Kairys (1979) design an algorithm to find the optimal number of strikes allowed under a slightly different definition of injustice. Ford (2010) examines the impact of varying the number of strikes allowed on conviction rates in a similar framework, without reference to an explicit definition of injustice. Feddersen and Pesendorfer (1998) consider the same types of injustice as Neilson and Winter (2000) in the context of strategic voting, but jurors in that context are ex ante identical, so there is no scope to examine aspects of jury selection.

⁶Shayo and Zussman (2011), Alesina and La Ferrara (2011), Abrams et al. (2011), and Iyengar (2011) provide evidence that the race of judges, victims, and defendants can affect trial outcomes in other settings. There are many other studies on race and criminal trial outcomes that rely on mock trials and case studies of a handful of trials at a time. A fairly recent and comprehensive review of such studies is given by Sommers and Ellsworth (2003), and a more in-depth and critical review of the earliest of these studies is given by Pfeifer (1990). Bowers et al. (2001) include 340 trials in their analysis, but are unable to adequately control for heterogeneity across these trials. They find that the death penalty is three times as likely for a case with a black defendant whose victim is white and for which the jury consists of five or more white male jurors than for similar cases with more mixed juries.

⁷Some studies attempt to draw inferences on attorney strategies in jury selection by examining the types of jurors removed by each attorney. For example, this method has been implemented by Anwar et al. (2012a) to identify attorney preferences over juror age; by Baldus et al. (2001) to identify attorney preferences over juror race, gender, and age; and, for civil trials, by Diamond et al. (2009) to identify attorney preferences over juror race, gender, age, and income. However, these studies do not explore how attorney behavior depends on varying institutional constraints and strategic concerns, which we focus on.

pursue such extensions theoretically, and we empirically confirm our model's predictions concerning what attorneys can accomplish.

In relation to the existing theoretical and empirical literature on attorney behavior in voir dire and its impact on verdicts, our work provides several new and important contributions. First, we make more realistic assumptions about the information available to attorneys. Brams and Davis (1978) assume that attorneys can observe juror predispositions, while Nielson and Winter (2000) assume that attorneys cannot observe predispositions but know that an observable juror characteristic is predictive of a stronger predisposition towards the prosecution. We assume instead that jurors are described by a twodimensional set of characteristics: one, such as race, which is always observable; and a second, namely a predisposition to one side or the other, which can potentially be imperfectly observed if attorneys are empowered. When attorneys cannot observe predispositions at all, they have no choice but to rely on the common-knowledge correlation between the observable characteristic and unobservable leanings. But when empowerment allows attorneys to learn about juror inclinations, they can use this partial information *in addition to* the information already afforded by the observable characteristic. Our model thus includes the Nielson and Winter framework as a special case, while, in the general specification, avoiding the unrealistic full-information assumption of Brams and Davis.

Second, relative to existing studies, we broaden our focus beyond peremptory challenges. Peremptory challenges allow attorneys to strike any potential juror without stating a reason, with very few exceptions. In contrast, successful challenges for cause must be accompanied by an acceptable argument that the targeted potential jurors have demonstrated an inability to be impartial. Previous studies only treat attorney decisions concerning which potential jurors to strike with a limited number of available peremptory challenges, assuming that all potential jurors for whom an inability to be impartial could be demonstrated have been removed via challenges for cause in an unmodeled earlier stage. Our modeling assumptions reflect a more realistic situation in which attorneys attempt to use both strikes for cause and peremptory strikes to remove jurors with unfavorable predispositions to their side. The ability of attorneys in our model to remove and object to the removal of potential jurors – and the effect of empowerment on this ability – is handled in a reduced-form manner that can incorporate strikes made via either type of challenge.

Third, we incorporate heterogeneity in attorney skill in our model, which follows naturally from these first two contributions. There is little potential role for attorney skill when juror predispositions are fully observable and only peremptory challenges are considered. However, the acquisition and interpretation of information, and persuasiveness in semantic arguments over what constitutes grounds for dismissal for cause, can reasonably be expected to depend on attorney facility. We hence allow for the possibility that more skilled attorneys can reap greater benefits from empowerment than less skilled attorneys in our model, and test for this possibility empirically.⁸

Finally, in our work, we consider juror predispositions directly, rather than trial verdicts. To be

⁸Abrams and Yoon (2007) and Shinall (2010) find that attorney skill is an important determinant of trial outcomes, but do not discuss this in the context of jury selection.

sure, the verdict is what all parties involved in the trial care about most. However, verdicts are a product of more than the predispositions of individual jurors. Of specific concern for our work is that verdicts are determined by jury deliberations, which can entail complex group dynamics and decision-making. Previous theoretical contributions skirt this issue by calculating the jury's overall probability of conviction as the simple product of each juror's probability of conviction, while noting the unsatisfactory nature of the implicit independence assumption. Instead, we take individual juror predispositions as the outcome variable of interest to attorneys. We consider this to be a desirable approach based on our postulation that attorneys are much better at assessing individual juror leanings than they are at predicting how the seated jurors will interact with one another in arriving at their final decision. Indeed, Gobert et al. (2009) suggest that techniques available to attorneys for assessing individual jurors are much more advanced than those for predicting deliberation dynamics, and that selecting "juries, not jurors" is still very much an emerging concept.

This paper is the most complete attempt to date to measure what attorneys can actually accomplish given varying degrees of empowerment in jury selection with regard to the multiple characteristics of the seated jurors they manage to retain. We find that skilled and empowered attorneys can successfully retain jurors favorably inclined to their side on seated juries, but that they do not necessarily accomplish this simply by striking according to juror race. In fact, the attorneys in our sample end up altering the racial composition of the seated jury relative to the jury pool very little: equivalent to no more than one juror of a given race in either direction on a twelve-person jury. The implication is that empowerment in voir dire can allow attorneys to uncover valuable information about jurors and therefore to avoid using racial stereotypes as a crutch. More generally, our findings provide strong evidence that extensive voir dire involving attorneys can lead to the seating of biased juries when opposing counsels are unequally skilled, yet the presence of this bias may not be necessarily detected in the observable characteristics of seated juries. We believe that our results could lead to more constructive policy advice than recommendations that have been made to restrict attorney freedoms in jury selection. We will leave specific discussion of policy implications for the conclusion, along with an attempt to address the important question of how our results might generalize to regions that we have not been able to include in our empirical analysis.

The rest of the paper will proceed as follows. Section 2 introduces the model and presents its main results, and Section 3 describes the dataset that we employ. Section 4 presents and discusses the main empirical findings and a number of robustness checks, and the following Section 5 provides some additional evidence corroborating our main results. We conclude by discussing some remaining questions and offering some policy suggestions in Sections 6 and 7.

2 A Model of Attorney Empowerment and Jury Selection

2.1 Voir Dire in Practice and in the Model

A very common organizational framework for voir dire employed with some variation in most American criminal trials is known as the strike-and-replace method.⁹ Under this framework, the "panel" or "venire" of potential jurors – which we can take as being approximately randomly drawn from the county population – is first randomly ordered. Sometimes the entire panel is initially examined at this stage, and strikes for cause exercised. In either case, the first several panelists from the random order are provisionally seated and potentially examined in more depth. Once the attorneys are satisfied that none of the provisionally seated jurors can be removed for cause, they exercise peremptory challenges in some pre-determined alternating sequence, usually with the prosecutor going first. Those provisionally seated jurors who are thus excused are then replaced by the panelists next in line. This cycle then repeats itself, sometimes with the restriction that seated jurors who have already been "passed" on by both attorneys have passed on, or when all peremptory challenges have been exhausted by both sides. In either case, the final seated jury is composed of the twelve jurors then seated, with the remaining venirepersons who have not already been struck also sent home.

The examination of potential jurors can be carried out primarily by the judge, by the judge with suggestions or a greater degree of participation from the attorneys, or primarily by the attorneys. Attorneys may also have access to the results of questionnaires filled in by potential jurors: either rudimentary surveys used by the court or, sometimes, "Supplementary Juror Questionnaires" (SJQs) that the attorneys have designed. The judge has full discretion over the decisions of how much attorney participation to allow and whether an SJQ is allowed. Individual judges tend to have strong views on how these decisions should be made and hold adamantly to them.

The primary concerns that are examined in any voir dire are the existence of personal relationships between panelists and any other parties involved in the trial, and the capacity to understand and contemplate the salient legal issues and evidence in a dispassionate and impartial manner. But many other subjects can also be explored, especially if attorneys have a high degree of participation. As voir dire questioning proceeds, attorneys challenge for cause by arguing that the responses of a given venireperson have revealed an inability to be impartial. The attorney on the opposite side is permitted to object to this argument and raise counter-arguments, and the judge must ultimately rule in favor of or against the strike. If the judge rules in favor of the strike, the potential juror is dismissed. Attorneys can strike an unlimited number of potential jurors for cause, as long as the judge can be sufficiently convinced that disqualification has been demonstrated.

Attorneys also have a limited number of peremptory strikes at their disposal. These allow the removal of potential jurors without the requirement that bias or any other deficiency be demonstrated.

⁹Our understanding of the voir dire process is based on Gobert et al. (2009) and Starr and McCormick (2001), as well as informal interviews with a number of legal professionals.

States and some counties provide guidelines on the number of peremptory challenges that should be made available, but judges are under no obligation to follow these guidelines, and exceptions are often made for various reasons. Nonetheless, the total number of peremptory strikes available is generally small relative to the size of the panel. There is also a chance that a peremptory strike can be disallowed: if an attorney suspects that the motivation for a strike is discriminatory in nature, a "Batson objection" can be raised. If the opposing attorney cannot respond by stating a non-discriminatory justification for the strike, it will be disallowed.

We model a simplified strike-and-replace voir dire procedure. The panel is assumed to be infinite, with a composition across all venireperson characteristics fixed and mirroring that of the population.¹⁰ The panel is arranged according to a random ordering that attorneys do not observe, and the single panelist at the top of this order is provisionally seated. If either attorney chooses to strike this panelist and is successful, the panelist is dismissed, and the next panelist from the random ordering is provisionally seated. This process continues until an attempted strike is unsuccessful or neither attorney chooses to attempt a strike. At that time, the provisionally-seated panelist becomes the sole member of the one-person seated jury that will hear the trial.¹¹

There is a probability that any strike attempt will be successful, and we allow this probability to differ by attorney and to depend on skill and empowerment. This is mostly meant to represent strikes for cause. With a certain probability, the panelist under examination will exhibit some traits that one of the attorneys can represent as demonstrating a lack of impartiality, at least to the point of rebutting any objections raised by the opposing counsel and convincing the judge. Empowered and skilled attorneys can better uncover such traits, and better argue over the semantics of how they reveal partiality. However, this framework can be slightly modified to incorporate peremptory strikes too. First of all, peremptory strikes can have a less-than-perfect success rate due to Batson objections. More generally, though, we could allow attorneys to have a much higher (or perfect) success rate for a limited number of strikes. This is an extension that we consider worthwhile and that we are currently pursuing. In the meantime, we propose that our present approach is capable of providing most of the insights that the more general approach will.

We will proceed for the rest of this section within this simplified framework. After formalizing some notation and concepts, we will clarify the roles of attorney empowerment and skill. Our main goal is to evaluate the probability that the seated juror will possess various attributes and characterize how this probability depends on attorney empowerment and skill.

¹⁰Courthouses can, in principle, bring in new panels for a given trial if the initially-drawn panel is exhausted by strikes, though this is rarely necessary. However, there have been concerns raised that panel composition does not accurately reflect population characteristics due to insufficient methods to construct and maintain master juror lists. Neither of these issues is important for our analysis.

¹¹The one-person jury model can be interpreted as a single round of decisions in a series of rounds in which attorneys decide how to fill a 12-person jury one juror at a time, which is essentially the approach taken by Brams and Davis (1978). The relaxation of this assumption, though cumbersome, is therefore straightforward.

2.2 Preliminaries

Suppose that an $\alpha \in (0,1)$ proportion of the panel is non-white and a $(1 - \alpha)$ proportion is white.¹² Each potential juror is described by two characteristics: his or her

- 1. race, $i = \{non white (n), white (w)\}$ and
- 2. standard required to convict, $s = \{high(h), low(l)\}$.

Race is fully observable to both attorneys, but standards required to convict are not. We interpret someone with a high standard required to convict – that is, someone requiring very strong evidence against the defendant to vote guilty – as someone with a predisposition or inclination towards the defense. We, therefore, refer to h- or high standard types as those with a predisposition towards the defense, and l- or low standard types as those with a predisposition towards the prosecution.

The defense attorney will always prefer seating an *h*-type to seating an *l*-type on the jury, and vice versa for the prosecution. $p_i \in (0, 1)$ is the proportion of jurors of race *i* who are predisposed to the defense (*h*-types), and the remaining $(1-p_i)$ is the fraction who are predisposed towards the prosecution (*l*-types). These proportions are common knowledge to both attorneys. We assume that the proportion of *h*-types is greater among non-whites than whites.

Assumption 1 (Race and Predispositions).

$$p_n > p_w. \tag{1}$$

Hence, without any additional information about the panelists, the defense will always prefer a nonwhite panelist over a white panelist, and the prosecution will always prefer to seat a white venireperson rather than a non-white.

Given the proportion of non-whites/whites and the fraction of their predisposition types in the potential juror population, we define

$$p = \alpha p_n + (1 - \alpha) p_w \tag{2}$$

as the proportion of panelists who are predisposed to favor the defense.

2.3 Roles of Attorney Empowerment and Skill

In voir dire, the judge may grant greater power to the attorneys, for example, by allowing them to participate to a greater degree in the examination of potential jurors or to use a Supplementary Juror Questionnaire. We assume that such empowerment has two potential effects. First, greater power in voir

 $^{^{12}}$ We will focus on race, because the issue of race has been at the forefront of policy discussion about potential biases in the process of jury selection. Moreover, focusing on one concrete observable characteristic allows for a more lucid exposition. However, note that all results hold for any observable characteristic.

dire may allow attorneys to find grounds to successfully strike any panelist with a higher probability than otherwise. We call this effect the "striking ability effect" of empowerment. Second, granting attorneys more power may allow them to acquire additional information about the predispositions of panelists and, thus, to better identify potential jurors who are more favorably-inclined to their side. We describe this effect as the "information effect" of empowerment.

We allow the magnitudes of both effects to depend on the relative skill levels of the attorneys. For simplicity of exposition, we assume that the defense attorney is more highly skilled than the prosecutor throughout our theoretical discussion. This assumption is partly motivated by the fact that it is common in criminal trials for the defense attorney to be more experienced than the prosecutor, a pattern also observed in our dataset. However, the model is symmetric from the perspective of each side, and our general predictions hold whether the defense or the prosecutor is more skilled.

In our theoretical discussion below, we examine the racial composition and the predispositions of the seated jury under varying institutional settings and conditions. First, we compare the race and the bias of the seated jury to the panel when attorneys are not empowered in voir dire. We then consider the impact of attorney empowerment on the seated jury when only the "striking ability effect" is operative, and when only the "information effect" of empowerment is at work. Finally, we assess the effect of empowerment when both effects are present.

2.4 No Attorney Empowerment

Let β be the baseline probability that a strike by either side is successful. We assume that greater rhetorical skill is ineffective without some information beyond what is revealed by a rudimentary judgeconducted examination of panelists. Thus, without empowerment, attorneys have no means to improve this success rate, regardless of any skill advantage.

With probability α , the first potential juror to be provisionally seated will be non-white. The prosecution will want to strike this panelist, since no information is available besides the panelist's observable race and the *a priori* knowledge that non-whites are more likely to be predisposed to the defense. With probability $(1-\beta)$, the strike will be unsuccessful, and the non-white panelist will become the seated juror. With probability β , the strike will be successful, and the next panelist in the random order will be provisionally seated. There is then a probability of $(1-\alpha)$ that this next panelist will be white. If the next panelist is white, the defense attorney will attempt to strike this provisionally-seated juror with a success probability of β . It is clear from the symmetry of the reasoning and the equality of striking success rates that attorneys will be powerless to alter the probability that the seated juror will be non-white beyond the probability that any randomly-drawn panelist is non-white, namely α . It follows directly that the expected probability that the seated juror will be predisposed towards the defense is just p, exactly equal to that proportion in the panel. To summarize, when attorneys are unempowered, regardless of attorney skill, the probability that the seated juror is non-white and predisposed towards the defense is the same as if the jury were randomly drawn from the panel.

2.5 Empowerment Increases Striking Ability Only

Assume now that attorney empowerment allows attorneys to find a broader basis for striking a panelist, thereby increasing the probability that an attempted strike will be successful. Let β_p be the prosecutor's striking success rate with attorney empowerment, and likewise, let β_d be the defense attorney's striking success rate when attorneys are empowered, where both of these are greater than the baseline striking success rate without empowerment. We further assume that the effect of empowerment on striking success is greater for the more highly skilled attorney, because attorneys with a greater rhetorical facility will be better able to support their claims of a lack of juror impartiality in the face of objections from their opponents. This assumption also allows for greater attorney skill to improve the chance of blocking an opponent's attempted strikes through more convincing counter-arguments. Since, for the sake of exposition, we are treating the defense attorney as the more highly skilled attorney, this implies that $\beta_d > \beta_p$. Combining these assumptions, we have

Assumption 2 (Empowerment and Striking Ability).

$$0 < \beta < \beta_p < \beta_d < 1. \tag{3}$$

The voir dire proceeds as follows. Each attorney decides whether to strike the panelist at the top of the random order or not. If the panelist is not struck or an attempted strike fails, then the panelist will be seated. If the panelist is struck successfully, he/she is replaced by the next individual from the top of the randomly-ordered venire, and the attorneys decide to attempt a strike or not (and argue over the basis for the strike). The process continues until a juror is successfully seated.

Let N be the probability that the panelist who is eventually seated is non-white. Additionally, let N_n be the probability that the seated juror will be non-white if a non-white panelist is at the head of the queue. N_w is, then, the probability that the seated juror will be non-white if a white panelist is at the head of the queue. Our assumption that the panel is of infinite size ensures that the identity of the juror at the head of the queue does not change the composition of the remaining panel.

With no additional information about the predispositions of the potential jurors aside from race, the defense (prosecuting) attorney will always try to strike when a white (non-white) panelist is at the head of the queue. Let's first consider the case in which a non-white panelist is at the head of the queue. With probability $(1 - \beta_p)$, the panelist will survive the prosecutor's strike request and end up as the seated juror. With probability β_p , he/she will be successfully struck and replaced by a random panelist at the head of the queue. Therefore, N_n can be defined as

$$N_n = (1 - \beta_p) + \beta_p N. \tag{4}$$

Similarly, when a white is at the head of the queue, he/she will survive the defense attorney's strike with probability $(1 - \beta_d)$, or with probability β_d , will be struck and be replaced by a random panelist

at the head of the queue. N_w is then

$$N_w = \beta_d N. \tag{5}$$

The overall expected probability that the panelist who is eventually seated is non-white is a weighted average of N_n and N_w , with weights given by the probability that a non-white will be at the head of the queue, α , versus a white, $(1 - \alpha)$, such that

$$N = \alpha N_n + (1 - \alpha) N_w. \tag{6}$$

This probability is then defined completely by the system of three equations (4), (5), and (6). Solving this system yields

$$N = \frac{\alpha(1 - \beta_p)}{(1 - \beta_d) + \alpha(\beta_d - \beta_p)}.$$
(7)

From (7), it is easy to see that $N > \alpha$ as long as $\beta_d > \beta_p$, which holds by Assumption 2.

It is also intuitive that the probability that the seated juror is predisposed to the defense – which we call P – is higher than that in the panel, p. This probability is defined as

$$P = p_n N + p_w (1 - N), (8)$$

which, after substitution of (7) for N, yields

$$P = \frac{\alpha(1-\beta_p)p_n + (1-\alpha)(1-\beta_d)p_w}{(1-\beta_d) + \alpha(\beta_d - \beta_p)}.$$
(9)

In (9), we can show that P > p as long as $\beta_d > \beta_p$ and $p_n > p_w$, which hold directly by Assumptions 1 and 2.

These results on the race and predisposition of the seated juror are intuitive. When empowerment increases striking ability only, the attorneys are simply able to more successfully strike on the basis of their *a priori* racial preferences; and when the defense attorney has a skill advantage, the defense is relatively more successful at striking than the prosecution, and so the outcome will more frequently benefit the defense attorney on net.

Proposition 1. When empowerment only increases attorney striking ability and the defense is more skilled,

- (i) the probability that the seated juror will be non-white is higher, and
- (ii) the probability that the seated juror will be predisposed towards the defense is also higher

than if the jury were randomly drawn from the panel.

2.6 Empowerment Increases Information Only

Now consider a setting in which empowerment does not increase the striking ability of attorneys (i.e. $\beta_p = \beta_d = \beta$) but allows attorneys to acquire additional information about the panelist and to better identify panelists who are favorably inclined to their side. Specifically, assume that when empowered, attorneys observe a dichotomous signal $\theta \in \{High(H), Low(L)\}$ that is correlated with the panelists' predispositions. For simplicity, we assume that only the more skillful attorney of the two sides is able to observe θ .¹³ A panelist of type *s* exhibits *H* with probability σ_s and exhibits *L* with probability $1 - \sigma_s$. We assume that the σ 's are independent of race and that $\sigma_h > \sigma_l$, so that any *h*-type panelist has a higher probability of exhibiting *H* than any *l*-type panelist. The probability that a panelist of race *i* exhibits *H* is then

$$\gamma_i = p_i \sigma_h + (1 - p_i) \sigma_l, \tag{10}$$

and, under our parameters assumptions about p_i and σ_h , we know that $\gamma_n > \gamma_w$. We can also express the conditional posterior probability that a juror of race *i* exhibiting *H* is actually an *h*-type as

$$\delta_{iH} = \Pr[s = h|i, \theta = H] = \frac{p_i \sigma_h}{p_i \sigma_h + (1 - p_i)\sigma_l}.$$
(11)

Similarly, the conditional posterior probability that a juror of race i and signal $\theta = L$ is actually an h-type is

$$\delta_{iL} = Pr[s = h|i, \theta = L] = \frac{p_i(1 - \sigma_h)}{p_i(1 - \sigma_h) + (1 - p_i)(1 - \sigma_l)}.$$
(12)

Because the less-skilled prosecutor does not see θ , his strategy in voir dire remains the same as before. That is, the prosecutor will try to strike non-white panelists and will attempt to block the striking of white panelists by the defense.¹⁴ However, the defense attorney, now having seen θ , updates her race-based *a priori* belief about any panelist's predisposition with information revealed by θ , according to (11) and (12).

Note that, given our our assumptions $p_n > p_w$ and $\sigma_h > \sigma_l$, a non-white panelist exhibiting H has the highest probability of being an h-type (δ_{nH}) , while a white panelist exhibiting L has the lowest probability of being an h-type (δ_{wL}) . However, it remains unclear whether $\delta_{wH} > \delta_{nL}$. This inequality will hold if and only if the following condition is met.

¹³One could also assume that both attorneys observe θ , but that the more skillful attorney can observe the signal more accurately. This alternative assumption, while more realistic, has not been employed here because it complicates the model without providing additional insight.

¹⁴In fact, we implicitly make a rather stronger set of assumptions about the skill-disadvantaged attorney in order to ensure that this behavior arises. In addition to not allowing the prosecutor to see the signal, we also assume that the prosecutor is unable to learn anything by observing the striking behavior of the defense attorney, or otherwise condition his own striking behavior on the anticipated or actual striking behavior of the defense attorney. The prosecutor thus behaves as though fully unaware of the potential existence of any signal. Allowing the less-skilled attorney to instead observe a less precise signal necessitates the relaxation of this package of assumptions, which can give rise to some anomalous cases.

Condition 1 $(\delta_{wH} > \delta_{nL})$.

$$\frac{\sigma_h}{\sigma_l} \frac{1 - \sigma_l}{1 - \sigma_h} > \frac{p_n}{p_w} \frac{1 - p_w}{1 - p_n}.$$
(13)

Condition 1 states that the odds ratio of exhibiting the signal H between h-types and l-types is greater than the odds ratio of being an h-type between non-whites and whites. Intuitively, $\delta_{wH} > \delta_{nL}$ – such that the defense attorney will prefer some white panelists to some non-white panelists – if and only if the signal θ is sufficiently more informative about juror predispositions than race alone.

2.6.1 Defense Attorney's Reservation Striking Threshold

When empowerment enables the more skilled attorney to better identify panelists predisposed to their side, opposing attorneys can now potentially "agree" on their preference to strike or to seat a panelist. For example, suppose the defense attorney is more skilled and can take advantage of the information effect of empowerment. The prosecution will always want to seat a white panelist as before empowerment. However, now the defense may also want to seat a white panelist if her posterior belief that the panelist is an *h*-type is sufficiently high. We, therefore, slightly modify our assumptions on empowerment's effect on attorneys' striking abilities by assuming that 1) if neither attorney wants to strike the panelist at the head of the queue, that individual will be seated on the jury; and 2) if both attorneys want to strike a panelist, that panelist will be struck with probability 1. If the prosecution and the defense disagree, then the probability of a successful strike is defined by β as before.

Given the primitive parameters of the model and the informativeness of signal θ , the defense attorney (assumed to be the more skilled attorney) will take the static behavior of the prosecutor as given and choose her own striking behavior to maximize the expected probability that the seated juror will be predisposed to the defense. The defense attorney's strategy can be summarized by the choice of a reservation threshold δ_R such that she will attempt to strike a panelist if $\delta_{i\theta} < \delta_R$ and argue against any attempted strikes otherwise. This problem can be expressed formally as choosing

$$\delta_R \in \operatorname*{arg\,max}_{\underline{\delta}} P(\underline{\delta}) \tag{14}$$

where $P(\underline{\delta})$ is the probability that the seated juror will be predisposed towards the defense. $P(\underline{\delta})$ is defined as a function of $P_{i\theta}$, the probability that the seated juror will be predisposed to the defense given that panelist of race *i* exhibiting a signal θ is at the head of the queue:

$$P(\underline{\delta}) = \alpha \gamma_n P_{nH}(\underline{\delta}) + \alpha (1 - \gamma_n) P_{nL}(\underline{\delta}) + (1 - \alpha) \gamma_w P_{wH}(\underline{\delta}) + (1 - \alpha) (1 - \gamma_w) P_{wL}(\underline{\delta}).$$
(15)

The overall expected probability that a seated juror will be predisposed towards the defense, P, for a given threshold $\underline{\delta}$, is a weighted average of $P_{i\theta}$ for each i and θ , with weights given by the population proportions of the four $i\theta$ groups.

 $P_{i\theta}$ depends on the probability of signal given the panelist's race and the attorneys' corresponding behavior given their choice of the threshold $\underline{\delta}$. Consider the case in which a non-white exhibiting signal H is at the head of the panel queue. Since a non-white with signal H has the highest probability of being predisposed to the defense, it is always optimal for the defense attorney to try to argue against any attempted strikes of a non-white with signal H. The prosecutor will always want to seat this panelist. With $(1 - \beta)$ probability, the defense's striking attempt will be denied and the non-white, H signal panelist with probability of δ_{nH} is being predisposed for the defense will be seated. With β probability, the strike attempt will be successful, and a random individual from the panel will move to the top of the queue. These strategies are reflected in the definition of P_{nH} in (16).

Non-White, H signal at head of queue:

$$P_{nH}(\underline{\delta}) = (1 - \beta)\delta_{nH} + \beta P(\underline{\delta}) \tag{16}$$

Similarly, a white panelist exhibiting L has the lowest probability of being favorable for the defense. Therefore, it is always optimal for the defense to try to strike a white with signal L with β probability of success while the prosecutor will argue against attempted strikes of any white. Therefore, P_{wL} can be defined by (17).

White, L signal at head of queue:

$$P_{wL}(\underline{\delta}) = (1 - \beta)\delta_{wL} + \beta P(\underline{\delta}) \tag{17}$$

The probabilities that the seated juror will be an *h*-type when a non-white exhibiting L or a white exhibiting H is at the head of the queue depend on the threshold $\underline{\delta}$ relative to the $\delta_{i\theta}$'s. For example, if a white exhibiting signal H is at the head of the queue, $\delta_{wH} \geq \underline{\delta}$ represents a case in which neither the defense nor the prosecution attempt to strike him. Then the expected probability that the seated juror will be an *h*-type when a white exhibiting H is at the head of the queue in (18) is exactly equal to δ_{wH} , since he will be seated with probability 1 given our assumptions about attorney striking success. However, if $\delta_{wH} < \underline{\delta}$, this represents a case in which the defense attempts to strike, and the panelist will hence only be seated with probability $(1-\beta)$ or will be replaced by a random panelist with probability β .

White, H signal at head of queue:

$$P_{wH}(\underline{\delta}) = \begin{cases} \delta_{wH} & \text{if } \delta_{wH} \ge \underline{\delta} \\ (1-\beta)\delta_{wH} + \beta P(\underline{\delta}) & \text{otherwise} \end{cases}$$
(18)

Finally, if a non-white exhibiting L is at the head of the queue as in (19), $\delta_{nL} < \underline{\delta}$, both the defense and the prosecution will attempt to strike him. In this case, he will be removed with certainty and replaced with a random panelist. However, if $\delta_{nL} \ge \underline{\delta}$, the panelist will be seated with probability

 $(1-\beta)$, because the defense will attempt to seat him while the prosecution will try to remove the panelist.

Non-White, L signal at head of queue:

$$P_{nL}(\underline{\delta}) = \begin{cases} (1-\beta)\delta_{nL} + \beta P(\underline{\delta}) & \text{if } \delta_{nL} \ge \underline{\delta} \\ P(\underline{\delta}) & \text{otherwise} \end{cases}$$
(19)

Therefore, the reservation threshold δ_R for the defense attorney is defined as the solution to (14) given (15) to (19). In general, there will be a range of $\underline{\delta}$ over which U is maximized; we do not need to characterize δ_R more precisely than noting that it belongs to this range.¹⁵ This probability is hence maximized when the defense attorney follows the strategy of attempting to strike any panelist with $\delta_{i\theta} < \delta_R$ and arguing against any attempted strikes otherwise.

2.6.2 Race and Predisposition of the Seated Juror

The impact of empowerment's information attorney on the race and predisposition of the seated jury depends on the informativeness of the signal θ and whether Condition 1 is met such that $\delta_{wH} > \delta_{nL}$. We will limit our discussion to two key cases regarding the striking behavior of the defense attorney and the strength of the signal.

Case 1: $\delta_{nH} > \delta_{nL} > \delta_R > \delta_{wH} > \delta_{wL}$

In this case, Condition 1 does not hold, so that the information revealed by the signal is not enough to overcome the defense attorney's *a priori* preference for non-whites. Further, non-whites represent a large enough fraction of the population and are likely enough to be predisposed to the defense that the defense attorney chooses to set her reservation threshold δ_R above δ_{wH} . The net result is that the defense attorney will attempt to strike any white panelist regardless of θ , and will object to attempts by the prosecutor to strike any non-white panelists, again regardless of θ . Therefore, this case is effectively the same as the no empowerment case. The probabilities that the seated juror will be non-white and the probability that the seated juror will be predisposed towards the defense are the same as if a randomly selected juror were seated.

Case 2: $\delta_{nH} > \delta_{wH} > \delta_R > \delta_{nL} > \delta_{wL}$

A more interesting case is when Condition 1 is met, so that the defense attorney is able to identify some white panelists with a higher probability, conditional on the signal, of being predisposed to the defense than some non-white panelists. Furthermore, assume that the fraction of the population exhibiting

¹⁵For example, one can define δ_R as the min{arg max $P(\underline{\delta})$ }.

signal H is moderate enough that it is optimal for the defense attorney to set her reservation striking threshold δ_R above δ_{nL} . Thus, the defense attorney will argue against the prosecutor's attempted strikes of non-white panelists exhibiting H, but will herself attempt to strike non-white panelists exhibiting L, joining in with the prosecutor's efforts in this regard. Furthermore, the defense attorney will not attempt to strike any white panelists exhibiting H, and since the prosecutor will never attempt to strike any white panelist, a white panelist exhibiting H that is at the top of the random order of panelists will become the seated juror without objection from either side.

Therefore, in this case, N, the expected probability that the panelist who eventually becomes the seated juror will be non-white, is defined by the following system of equations:

$$N_{nH} = (1 - \beta) + \beta N$$

$$N_{nL} = N$$

$$N_{wH} = 0$$

$$N_{wL} = \beta N$$

$$N = \alpha \gamma_n N_{nH} + \alpha (1 - \gamma_n) N_{nL} + (1 - \alpha) \gamma_w N_{wH} + (1 - \alpha) (1 - \gamma_w) N_{wL}.$$

Solving for N, we have

$$N = \frac{\alpha \gamma_n (1 - \beta)}{1 - \alpha [1 - \gamma_n (1 - \beta)] - (1 - \alpha) [(1 - \gamma_w) \beta]}.$$
 (20)

It can be shown that $N < \alpha$ always, as long as Condition 1 and our basic parameter assumptions hold.¹⁶ Therefore, in this case, the probability that the seated juror will be non-white is *less* than if the jury were chosen randomly.

In summary, whenever the attorneys differ in their desire to strike a panelist, the panelist will be struck successfully with probability β on net. The attorneys always disagree on white panelists exhibiting L and on non-white panelists exhibiting H. However, the attorneys can potentially agree on the two intermediate groups, depending on the quality of the signal and the defense attorney's reservation striking threshold. The result is either that white panelists exhibiting H will be struck with probability zero (rather than β); non-white panelists exhibiting L will be struck with probability one (rather than β); or both.¹⁷ The net effect of any of these, in combination with the fixed and common probability of a panelist from the extreme groups being struck, is to reduce the probability that a white

¹⁶The demonstration of this claim is tedious, as the expression involves all of the parameters of the model through the definition of γ_i in equation (10). It is more straightforward to show a sufficient but not necessary condition in terms of the striking success rates of each attorney. This sufficient condition is $\beta > p_n/p_w \times (1 - \beta)$. Intuitively, this condition is saying that it must be sufficiently more likely that the defense attorney successfully strikes a white panelist exhibiting L (hence obtaining another draw from the panel) than it is for the prosecutor to fail to strike a non-white panelist exhibiting H (which would result in the voir dire terminating with a non-white seated juror). This condition is fairly stringent for many parameter values. However, as noted, the only necessary conditions are the baseline parameter assumptions we have made.

¹⁷The less skilled prosecutor does not observe the signal, as per our assumption stated earlier, but rather "agrees" with the defense attorney in these cases by virtue of his usual strategy of only attempting to strike non-whites.

panelist will be struck relative to that of a non-white panelist being struck.

We now turn to examining the impact of the information effect of empowerment on the prejudice of the seated jury. The most straightforward way to calculate P is to use the relevant forms of equations (15) through (19):

$$P_{nH} = (1 - \beta)\delta_{nH} + \beta P$$

$$P_{nL} = P$$

$$P_{wH} = \delta_{wH}$$

$$P_{wL} = (1 - \beta)\delta_{wL} + \beta P$$

$$P = \alpha\gamma_n P_{nH} + \alpha(1 - \gamma_n)P_{nL} + (1 - \alpha)\gamma_w P_{wH} + (1 - \alpha)(1 - \gamma_w)P_{wL}.$$

Solving for P yields

$$P = \frac{\alpha \gamma_n (1-\beta) \delta_{nH} + (1-\alpha) \gamma_w \delta_{wH} + (1-\alpha) (1-\gamma_w) (1-\beta) \delta_{wL}}{1-\alpha [1-\gamma_n (1-\beta)] - (1-\alpha) [(1-\gamma_w)\beta]}$$
(21)

where it can be shown that P > p under our parameter assumptions, meaning that the probability of that the seated juror will be predisposed towards the defense is higher than in the panel overall.¹⁸ Therefore, we have the following proposition when attorney empowerment increases the information available to the more skilled attorney without increasing his/her striking ability.

Proposition 2. When empowerment only allows attorneys to better identify panelists' predispositions, and the defense attorney is more skilled,

- (i) the probability that the seated juror will be non-white is lower, and
- (ii) the probability that the seated juror will be predisposed towards the defense is higher

than if the juror were randomly drawn from the panel.

2.7 Empowerment Increases Striking Ability and Information

Finally, we now turn to the case in which empowerment not only increases the striking ability of the attorneys, but also allows attorneys to acquire additional information about the panelist and to better identify panelists who are favorably-inclined to their side.

¹⁸Again, this claim is tedious to demonstrate algebraically. However, it is clear that it must hold, since this is the objective that the defense attorney is attempting to maximize. If P were less than p, the defense attorney would change striking behavior – for example, to that of Case 1 above – and be able to obtain a different P > p.

2.7.1 Defense Attorney's Reservation Striking Threshold

We continue to assume that if neither attorney wants to strike the panelist at the head of the queue, that individual will be seated on the jury, and if both attorneys want to strike a panelist, that panelist will be struck with probability 1. As before, if the prosecution and the defense disagree, then the probabilities of successful strikes are defined by β_p and β_d , respectively. Under the expositional assumption that the defense attorney is more skilled, we again impose the condition in Assumption 2 that the striking success rate for the defense β_d is higher than for the prosecution β_p .

As in the last setting, given the primitive parameters of the model and the informativeness of θ , the defense attorney will take the static behavior of the prosecutor as given, and choose her own striking behavior to maximize the expected probability that the seated juror will be predisposed to the defense. Similar to when empowerment only had an information effect, we can formally describe the defense attorney's choice of the reservation threshold δ_R .

$$\delta_R \in \arg\max_{\underline{\delta}} P(\underline{\delta}) \tag{22}$$

where $P(\underline{\delta})$ is defined as before in equation (15), but

$$P_{nH}(\underline{\delta}) = (1 - \beta_p)\delta_{nH} + \beta_p P(\underline{\delta}), \qquad (23)$$

$$P_{wL}(\underline{\delta}) = (1 - \beta_d)\delta_{wL} + \beta_d P(\underline{\delta}), \qquad (24)$$

$$P_{nL}(\underline{\delta}) = \begin{cases} (1 - \beta_p)\delta_{nL} + \beta_p P(\underline{\delta}) & \text{if } \delta_{nL} \ge \underline{\delta} \\ P(\underline{\delta}) & \text{otherwise,} \end{cases}$$
(25)

and

$$P_{wH}(\underline{\delta}) = \begin{cases} \delta_{wH} & \text{if } \delta_{wH} \ge \underline{\delta} \\ (1 - \beta_d)\delta_{wH} + \beta_d P(\underline{\delta}) & \text{otherwise.} \end{cases},$$
(26)

with striking success rates of β_d and β_p .

Since a non-white exhibiting H has the highest probability of being predisposed to the defense and a white exhibiting L has the lowest, it is always optimal for the defense attorney to try to argue against attempted strikes of a non-white H and to attempt to strike a white L. The less-skilled prosecutor will always want to strike any non-white and argue against attempted strikes of any white, as explained above, and the defense attorney anticipates this fully. These choices are reflected in equations (23) and (24) defining P_{nH} – the expected probability that the seated juror will be predisposed to the defense when a non-white exhibiting H is at the head of the queue – and P_{wL} – the corresponding probability when a white exhibiting L is at the top of the order – respectively.

The probabilities that the seated juror will be an h-type when a non-white exhibiting L or a white

exhibiting H is at the head of the queue depend on the threshold $\underline{\delta}$ relative to the $\delta_{i\theta}$'s. For example, if a white exhibiting signal H is at the head of the queue, $\delta_{wH} \geq \underline{\delta}$ represents a case in which neither the defense nor the prosecution attempt to strike him. Then the expected probability that the seated juror will be an h-type when a white exhibiting H is at the head of the queue in (26) is exactly equal to δ_{wH} , since he will be seated with probability 1 given our assumptions about attorney striking success. However, if $\delta_{wH} < \underline{\delta}$, this represents a case in which the defense attempts to strike, and the panelist will hence only be seated with probability $(1 - \beta_d)$, or will be replaced by a random panelist with probability β_d .

Finally, the overall expected probability that a seated juror will be predisposed towards the defense, P in (15) for a given threshold $\underline{\delta}$, is a weighted average of (23) through (26), with weights given by the population proportions of the four $i\theta$ groups.

2.7.2 Race and Predisposition of the Seated Juror

As in the setting with only information effect of empowerment, the impact of empowerment's information and striking ability on the race and predisposition of the seated jury depends on the additional informativeness of the signal θ .

Case 1: $\delta_{nH} > \delta_{nL} > \delta_R > \delta_{wH} > \delta_{wL}$

Condition 1 does not hold, and that the information revealed by the signal is not enough to overcome the defense attorney's *a priori* preference for non-whites. Moreover, non-whites represent a large enough fraction of the population and are likely enough to be predisposed to the defense that the defense attorney chooses to set her reservation threshold δ_R above δ_{wH} . The net result is that the defense attorney will attempt to strike any white panelist, regardless of θ , and will object to attempts by the prosecutor to strike any non-white panelists, again regardless of θ . Therefore, this case is effectively the same as when empowerment only increased the striking abilities of the attorneys. The probability that the seated juror will be non-white is *higher* than if a randomly selected juror were seated, as is the probability that the seated juror will be predisposed towards the defense.

Case 2: $\delta_{nH} > \delta_{wH} > \delta_R > \delta_{nL} > \delta_{wL}$

With Condition 1 met, the defense attorney is able to identify some white panelists with a higher posterior probability of being predisposed to the defense than some non-white panelists. Furthermore, the fraction of the population exhibiting signal H is moderate enough that it is optimal for the defense attorney to set her reservation striking threshold δ_R above δ_{nL} . Thus, the defense attorney will argue against the prosecutor's attempted strikes of non-white panelists exhibiting H but will attempt to strike non-white panelists exhibiting L, joining in with the prosecutor's efforts. Moreover, the defense attorney will not attempt to strike any white panelists exhibiting H, and since the prosecutor will never attempt to strike any white panelist, a white panelist exhibiting H that is at the top of the random order of panelists will become the seated juror without objection from either side. Therefore, in this case, N, the expected probability that the panelist who eventually becomes the seated juror will be non-white, is defined by the following system of equations:

$$N_{nH} = (1 - \beta_p) + \beta_p V$$

$$N_{nL} = V$$

$$N_{wH} = 0$$

$$N_{wL} = \beta_d V$$

$$N = \alpha \gamma_n V_{nH} + \alpha (1 - \gamma_n) V_{nL} + (1 - \alpha) \gamma_w V_{wH} + (1 - \alpha) (1 - \gamma_w) V_{wL}$$

Solving for V, we have

$$N = \frac{\alpha \gamma_n (1 - \beta_p)}{1 - \alpha [1 - \gamma_n (1 - \beta_p)] - (1 - \alpha) [(1 - \gamma_w) \beta_d]}.$$
(27)

It can be shown that $N < \alpha$ always, as long as Condition 1 and our basic parameter assumptions hold.¹⁹ Therefore, the probability that the seated juror will be non-white is *less* in this case than if the jury were chosen randomly.

However, as in the previous case, P, the probability that the seated juror will be predisposed towards the defense, is higher than in the panel overall. Using the specific form of equations (23) through (26) for this case, which are

$$P_{nH} = (1 - \beta_p)\delta_{nH} + \beta_p P$$
$$P_{nL} = P$$
$$P_{wH} = \delta_{wH}$$
$$P_{wL} = (1 - \beta_d)\delta_{wL} + \beta_d P$$

we can solve for P:

$$P = \frac{\alpha \gamma_n (1 - \beta_p) \delta_{nH} + (1 - \alpha) \gamma_w \delta_{wH} + (1 - \alpha) (1 - \gamma_w) (1 - \beta_d) \delta_{wL}}{1 - \alpha [1 - \gamma_n (1 - \beta_p)] - (1 - \alpha) [(1 - \gamma_w) \beta_d].}$$
(28)

It can be shown that P > p under our parameter assumptions. In Case 2, we have thus shown an example for which empowerment that increases attorney striking abilities *and* allows them to more

¹⁹Intuitively, a sufficient condition requires that it is sufficiently more likely that the defense attorney successfully strikes a white panelist exhibiting L (hence obtaining another draw from the panel) than it is for the prosecutor to fail to strike a non-white panelist exhibiting H (which would result in the voir dire terminating with a non-white seated juror). This condition is fairly stringent for many parameter values. However, as noted, the only necessary conditions are the baseline parameter assumptions we have made.

precisely identify juror predispositions in this case leads to a *lower* probability that the seated juror will be non-white, compared to the proportion of non-whites in the population.

Other Cases

Cases 1 and 2 are the most extreme of the many possible cases. Others are nonetheless interesting. When the proportion of the panel exhibiting the High signal is small, the defense attorney may choose to refrain from striking all but white panelists exhibiting L, achieving a probability that the seated juror is non-white slightly greater than α . On the other hand, when this proportion is large, the defense attorney may wish to attempt to strike all but non-white panelists exhibiting H; depending on the specific value of σ_h , the probability that the seated juror is non-white can be slightly less than or slightly greater than α . We will avoid a detailed exposition of these cases for the present, as Cases 1 and 2 are sufficient to support the following Proposition.

Proposition 3. When empowerment increases attorney striking ability and allows attorneys to better identify panelists' predispositions, and the defense attorney is more skilled,

- (i) the probability that the seated juror will be non-white may be higher or lower, yet
- (ii) the probability that the seated juror will be predisposed towards the defense is always higher

than if the juror were randomly drawn from the panel.

2.8 Summary of Empirical Predictions

Comparing Propositions 1 through 3 to when attorneys are unempowered, the model produces a clear and intuitive testable prediction. When the defense attorney is relatively more skilled than the prosecutor, attorney empowerment in voir dire is associated with an increase in the probability that the seated juror will be predisposed to the defense. However, predictions concerning the race of the seated juror are less clear. If the nature of attorney empowerment is only such that it improves the striking ability of attorneys, empowerment will be associated with an *increase* in the probability that the seated juror will be non-white. In direct contrast, if the nature of attorney empowerment is only such that it allows attorneys to learn more about juror predispositions, empowerment will be associated with a *decrease* in the probability that the seated juror will be non-white. And, therefore, if both mechanisms are operative, empowerment could be associated with either an increase or a decrease in the probability that the seated juror will be non-white. Since the precise nature of attorney empowerment and of any information uncovered by attorneys in voir dire will be unobserved to us in our empirical analysis, we will not be able to test these predictions concerning race directly. Instead, our approach will be to attempt to draw inferences about the nature of empowerment from our results. Before presenting these results, we describe our dataset.

3 The NCSC/ICPSR Hung Juries Dataset

Our empirical analysis relies on a detailed examination of 351 felony jury trials in four major county courts across the U.S. collected by Hannaford-Agor et al. (2002, 2003) for the National Center for State Courts (NCSC) during 2000 and 2001 and disseminated by the Interuniversity Consortium for Political and Social Research (ICPSR). The four courts – Los Angeles County Superior Court in California, Maricopa County Superior Court in Arizona, Bronx County Supreme Court in New York, and District of Columbia Superior Court in Washington, D.C. – were chosen based on their high volume of felony jury trials and their willingness to cooperate with the data collection process and guidelines. Although the NCSC's main goal in the study was to provide an empirical evaluation of hung juries, the dataset is not limited to trials ending in a hung jury, and indeed includes all non-capital felony trials held in these courts during the specified periods of data collection.²⁰

The NCSC study provides a comprehensive look at the trial and all the parties involved: the defendant(s) and the victim(s) (if any), the judge, the attorneys, and, most importantly for our analysis, the seated jurors. We do not have information about the initial jury pool from which the seated jury was formed, except for the initial panel size.²¹ The case data in the NCSC study provide researchers with an extensive look at all the charges, the race and sex of the defendant(s) and victim(s), and the voir dire process that led to the seated jury. The NCSC questionnaires also asked the presiding judge of each trial for his/her evaluation of the evidence, case complexities, and attorney skills. The NCSC data also contain a description of the dynamics of each juror's opinion formation, and of pre- and post-deliberation perceptions and opinions regarding the defendant(s) and victim(s). To our knowledge, the richness of the jury demographic and opinion information available in the NCSC study is unmatched by any other readily available dataset.

Although the NCSC dataset contains information on 351 cases and 3,497 jurors in total, not every individual in our data answered every survey question. In our main specifications, which use responses from the case, judge, and juror modules of the dataset, we have data on 1,883 jurors across 256 trials. The attorney module has the lowest response rate, and in some specifications with attorney controls, we have as few as 142 trials represented.²² All data were collected confidentially, and no information was made available to any of the trial participants during the trial or during any subsequent appeals.²³

²³These were conditions of the data collection agreements. We do not observe if any of the trials in the dataset have

²⁰Each county had a single period of data collection ranging from 4 to 11 months in length, with data collection ending in Los Angeles in October 2000 before beginning elsewhere, though with the other three periods overlapping in various months in 2001. D.C. contributes the largest number of trials to the dataset, though each court is fairly evenly represented. The death penalty was available in all counties except for D.C. at the time of data collection, though the distribution of charges in the dataset is roughly the same in each county.

²¹We can, however, infer the average characteristics of the panel from those of the county as a whole, since, as was discussed previously, a panel is essentially a random sample of the county population. We include county fixed effects in all of our empirical specifications, in part to account for the composition of the panel as much as possible.

²²There are no large differences in most of the case characteristics between the trials in and out of our estimation sample. Compared to cases that are not in our main estimation sample, trials in our sample are slightly more likely to be from Maricopa, less likely to have high attorney participation, more likely to have a higher skilled defense attorney in the opinion of the judge, and more likely to involve a murder or manslaughter (and less likely to involve drug-related crimes). These results are available upon request

3.1 Variables of Interest

Below, we provide a detailed description of the empirical analogues we propose for some of the important theoretical variables from the model and provide an extensive discussion of the some of the relevant strengths and shortcomings of the data.

3.1.1 Juror Predispositions

Each juror was asked "Before you began deliberating with your fellow jurors at the end of the trial (after all of the evidence and the judge's instructions had been presented), which side did you favor?", which was answered on a scale of 1 ("Strongly favored the prosecution") to 7 ("Strongly favored the defense"). We interpret higher responses on this seven-point scale as manifesting greater doubts on the part of a juror that sufficiently incriminating evidence had been presented in relation to that juror's personal (and otherwise unobservable) standard required to convict. Thus, jurors with higher responses to this question are assumed to have entered deliberations with a general mindset more inclined towards the defense.

This variable closely resembles the concept of juror predispositions as it was conceived and employed in the construction of the model. Moreover, it is a reasonable objective that attorneys would want to target in practice. The wording of the question can encompass subtle juror leanings deriving potentially from common life experiences with the defendant, the resonance of certain pieces of evidence and argument, and so on. These are leanings that attorneys could conceivably predict in jury selection and to which they can appeal during the trial.

Jurors were also asked their general opinions of the defendant and any victims outside the context of the facts of the case, as well as a series of questions concerning the evolution of their opinions during and after deliberations. We feel that these questions are less accurate representations both of our notion of juror predispositions and of traits that attorneys realistically target. Jurors could very likely find the defendant sincere and the victim untrustworthy and less believable, yet still lean towards believing the defendant guilty. Furthermore, attorney manuals have only very recently begun to espouse the idea that attorneys should "select juries, not jurors," and are more likely to advise that deliberations are akin to a black box, so that that the surer route to success is to focus on individual jurors during jury selection, rather than attempt to predict deliberation dynamics. This traditional view goes back to Clarence Darrow, who is said to have remarked that the outcome of a trial has virtually been decided by the time the jury has been sworn in.

All juror questionnaires were distributed at the conclusion of the trial. Therefore, there exists a potential concern that this question on the side favored prior to deliberation is an inaccurate representation of jurors' actual states of mind at that time. For example, if the final verdict was the most salient memory for jurors at the conclusion of the trial, it is possible that they engaged subconsciously in *ex post* rationalization in answering questions regarding earlier stages. Although we do not have

since been retried or heard by an appeals court.

a way of satisfactorily ruling out this possibility, we find reassurance in the care taken by the survey designers and in the fact that this section of the survey was carefully and transparently constructed to elicit information on the sequential nature of the opinion formation of jurors over the entire trial and deliberations. For example, the survey includes several reminders to respondents to consider precise times during which jurors changed their opinions.

3.1.2 Attorney Empowerment

The main case data contains an indicator for whether the voir dire was conducted by the judge alone, the judge with questions suggested by attorneys, the judge and attorneys together, or primarily attorneys. We combine the first two and the latter two categories to form a dummy variable for whether attorneys had high participation or low participation. The main case data also contains a dummy variable for whether or not there was a Supplemental Juror Questionnaire (SJQ) used by the attorneys.²⁴ However, there is no other information available concerning SJQs. For example, we do not observe if it was requested by one side or the other if present; we do not know if a request was made and denied if not present; we do not observe if both sides had access to the responses if present; and we do not observe any of the specific information that it contained if present.

Two important points should be noted concerning these two variables. First, it is implicitly assumed that each side is equally empowered by either instrument (high participation or a questionnaire) in terms of the additional information about jurors that the defense and prosecuting attorneys can obtain with them. This is in line with the assumptions of our model. There is no information available concerning which attorney was more active in questioning or how any SJQs were used, so there does not appear to be any feasible alternative to making this uniformity assumption. As in the model, however, attorney skill will be examined as a way of identifying heterogeneity across sides in the use that was made of these freedoms when they were available.

Second, it seems plausible to assume that the presence (or lack thereof) of these attorney freedoms is exogenous to other aspects of the trial. Individual judges have wide latitude in determining how the jury selection process is conducted, and most judges strictly adhere to a set of personal preferences, especially in regard to the degree of participation afforded to attorneys in questioning potential jurors. Judges can occasionally be persuaded to change policies if an attorney on either side is sufficiently aware of the judge's usual policies and submits a motion for a more extensive process than is typically allowed. This seems to be very rare in the case of attorney participation, but perhaps not unheard of in the case of supplementary questionnaires.

Without this exogeneity, our estimation would suffer a severe shortcoming. If, for example, judges always allowed high attorney participation in trials for crimes that had received extensive negative media attention, high participation could be highly correlated with jurors leaning away from the defense (based in part on preconceived notions of the trial that the voir dire did not remove). But this correla-

²⁴The survey instrument is in no way related to whether SJQs may or may not have been used in any trials, and the attorneys did not have access to any of the information from the survey instrument at any stage of the trial.

tion would be driven by the fact that both high participation and unfavorable opinions of the defendant are associated with the (unobserved in the dataset) media attention. If there were some degree of endogeneity of this sort (for example, if, for some trials, a questionnaire were allowed based on the motions of a defense attorney desiring as much information as possible on a venire of potential jurors suspected of holding extreme prejudices against the defendant), the estimated effects of attorney empowerment on juror predispositions would be biased in a negative direction. In Table 2 in the following section, we provide supporting evidence for the exogeneity of empowerment in relation to case-level variables used in our empirical analysis.

For most of our specifications, we use a single dummy variable for whether or not attorneys were empowered in voir dire. We construct this variable to take a value of 1 *either* if attorneys had high participation or if there was an SJQ in use, and to take a value of 0 if attorneys had neither freedom. However, we will also explore the robustness of our results to alternative definitions of attorney empowerment.²⁵

The dataset also contains additional information about the voir dire process. We observe the number of peremptory challenges actually exercised by each side, but not how many were available nor how this was determined. We also observe the total number of strikes for cause, but not how many suggested strikes for cause were objected to or denied. There is a dummy variable indicating whether any Batson objections were raised during voir dire, but if there were, we do not observe how many there were, which side raised them, whether they resulted in the finding of any actual violation, or what remedies were imposed, if any. There is also a dummy variable for whether the jury was anonymous, but there is no indication of whether this anonymity was from the perspective of record-keeping only or if the jury was literally empaneled in a separate room and voir dire conducted remotely. We observe the length of time voir dire took in hours, but not whether there was a time limit in place. Finally, we observe the total size of the initial panel. We have concerns about the completeness of the information furnished by these variables and about their exogeneity, but we do make use of them in some specifications.

3.1.3 Attorney Skill

The judge and each juror were asked separate questions of the form "How skillful was the [prosecutor/defense attorney] during the trial?", which were answered on a scale of 1 ("Not at all skillful") to 7 ("Very skillful").²⁶ To closely mirror the setup of the model, we would like to have an indicator for which attorney was the relatively more skilled. It would seem uncontroversial to assume that the judge

²⁵Two small idiosyncracies of the dataset should be mentioned briefly. All of the trials taking place in Bronx County in the dataset had high attorney participation in questioning. On the other hand, none of the trials in the dataset taking place in Maricopa had a questionnaire available. All of our specifications control for the county in which the trial took place, and we do not otherwise foresee these issues presenting problems within our empirical strategy, which is more focused on the interaction of attorney skill and empowerment. Estimations without trials in Bronx County show similar magnitudes of empowerment effects, albeit with less precision due to the loss of observations in the estimation sample.

²⁶Attorneys were also asked similar questions about their own and their opponent's skill. We do not make use of these variables due to concerns about subjectivity and for the pragmatic reason, as mentioned earlier, that attorneys had the lowest response rates.

would provide the most objective assessment of attorney skill, so that an attractive option would be to construct a dummy variable for whether or not the judge gave one attorney a higher skill rating than the other. However, the judge assigned the same skill rating to both attorneys in about half the cases.

In our empirical analysis, we provide two main proxies for attorney skill (and their differences) in a trial. First, to gain the most statistical power in our analysis of impact of empowerment by attorney skill, we construct a hybrid skill indicator for each attorney by calculating an average of the judge's rating and the mean juror rating in the same trial excluding the juror's own rating. We then calculate a relative skill index by subtracting the hybrid skill rating for the prosecutor from the hybrid skill rating for the defense attorney. Finally, we construct a dummy variable that we refer to as "Defense More Skilled" that takes a value of 1 when this relative skill index is greater than zero. We are aware of the somewhat arbitrary nature and potential shortcomings of this measure of relative attorney skill. For example, if jurors tend to rate the defense attorney more highly when they also recall being favorable to the defense, the jury average skill rating will be slightly positively correlated with a juror's own favorability response.²⁷ By excluding the juror's own ratings in calculating the mean juror rating at the trial level, we attempt to minimize the endogeneity concerns associated with this hybrid skill rating. Further, in reality, the opposing counsels' skills are probably not exactly equal to one another as indicated by the judge who may be constrained by a self-imposed pressure to appear impartial. Therefore, one can interpret the hybrid measure of attorney skill as a reasonable method of breaking this tie and getting a more precise measure of relative attorney skill within a trial.

Second, to demonstrate that our judge plus juror hybrid measure of attorney skill is not significantly biased, we report results that rely only on the judge's skill ratings of attorneys to determine the relative skill positions of the defense attorney and the prosecutor within a trial. Generally, our estimates show similar magnitudes of the empowerment effect as when using the hybrid measure of attorney skill, but they are less precisely measured.²⁸ The small loss of statistical power arises mainly from the fact that we define three categories of relative attorney skill when using the judge's rating only: defense attorney more skilled, and no skill difference.

Finally, a conceptual shortcoming of any of these skill measures is that they refer to skill in the trial arena as a whole, whereas, it could be argued, our model suggests that we should be focusing on attorney skill in voir dire exclusively. We respond to this, first of all, by noting that skill in various aspects of litigation are likely to be highly correlated, so that one lawyer displaying more skill than another during a trial as a whole should be a good proxy for that same lawyer having shown superior skill in any individual stage of the trial. However, the more important point is that we do not necessarily want to disregard the possibility that attorney skill in argument and presentation of evidence could affect individual juror predispositions, nor that attorneys could anticipate this during jury selection. This is

²⁷This assumes that the self-reported predisposition of jurors reflects some ex-post rationalization of their concluding sentiments about the case. However, noted in Section 3.1.1, we believe that this possibility is minimized by the care taken by the survey designers and in the fact that this section of the survey was carefully and transparently constructed to elicit information on the sequential nature of the opinion formation of jurors over the entire trial and deliberations.

²⁸We discuss a number of alternative measures of attorney skills in the results section.

neither ruled out by a more general interpretation of our theoretical framework nor by our empirical measurement of predispositions discussed above.

3.1.4 Other Variables

An important control variable in our analysis is the objective strength of evidence. For each trial, the judge was asked how close the case was based on the evidence that had been presented. The question was required to be answered on a scale of 1 ("Evidence strongly favored prosecution") to 7 ("Evidence strongly favored defense").²⁹

For juror race, we focus on the broad white/non-white dichotomous distinction that we used to illustrate the model. We recognize that the more specific black/white distinction typically receives the greatest focus in discussions of public policy, but the more salient difference across our dataset as a whole is the broader one. The non-white category is predictably heterogeneous across the four counties, but the constituent members in each case are, in general, assumed by commentators and legal practitioners to hold greater sympathies for defendants than their white counterparts. Other juror-level characteristics we can observe include education, income, age, gender, and religiousness. These are encoded categorically in the dataset, and we have converted them to dummy variables in some cases and continuous variables in other cases.³⁰

As has been mentioned before, we include county-level fixed effects in all of our specifications. We can also control for some other trial-level variates. The dataset provides a fairly detailed breakdown of trials by type of charge, which we coarsen into broad headings of violent, property and drug crimes. We can also control for defendant race and for the race and gender of any victims if present.³¹ Finally, we can control for whether or not the defendant had counsel provided at no cost (i.e. public defender), which we interpret as a rough indicator of the defendant's financial means.

3.2 Descriptive Statistics

Table 1 presents selected summary statistics of key variables in our dataset. The sample means are calculated for the observations in our main estimation sample of 256 trials. On average, jurors in our sample were mildly predisposed towards the prosecution prior to deliberations (comparing the mean response of 3.5 to the midpoint of 4 on the seven-point scale, with 7 representing "Strongly favored the defense"). However, individual responses are dispersed widely around this sample mean, and there are

 $^{^{29}}$ In an another specification, two dummy variables were been created from this variable. The first represents evidence strongly favoring the defense, and corresponds to responses of 6 or 7 (versus responses of 1 through 5). The second represents evidence strongly favoring the prosecution, and corresponds to responses of 1 or 2 (versus responses of 3 through 7). The overall excluded category therefore corresponds to a close trial in terms of the evidence favoring neither side very strongly. As this approach does not affect our results, we have chosen to retain the full scale.

³⁰In other specifications not reported in the current paper, we have tried controlling for additional juror characteristics, including occupation, broader income categories, and whether the juror is a foreperson. Addition of these controls do not effect our results. However, we lose an insignificant number of observations.

 $^{^{31}}$ We can also observe the defendant's gender, but we do not control for this because less than 10% of trials in the dataset had a female defendant. A handful of trials in our estimation samples have multiple defendants, but always of the same race.

	Mean	Std. Dev.	N	
Juror Predisposition (1 (pros.) - 7 (def.))	3.541	2.069	1883	(jurors)
Juror Non-White	0.520		1883	
Bronx	0.842		310	
LA	0.626		490	
DC	0.510		628	
Maricopa	0.202		455	
High Participation	0.621	0.486	256	(trials)
Questionnaire	0.203	0.403	256	
Either	0.688	0.464	256	
Both	0.137	0.344	256	
Defense Skill (Judge: 1 to 7)	5.102	1.433	256	
Defense Skill (Juror Average: 1 to 7)	4.419	1.126	256	
Prosecutor Skill (Judge: 1 to 7)	4.863	1.534	256	
Prosecutor Skill (Juror Average: 1 to 7)	4.781	1.067	256	
Defense More Skilled (Judge + Juror)	0.457	0.499	256	
Defense More Skilled (Judge Only)	0.313	0.464	256	
Prosecutor More Skilled (Judge Only)	0.211	0.409	256	
Crime: Murder and Manslaughter	0.156		256	
Crime: Sexual Crimes	0.055		256	
Crime: Robbery, Burglary, Arson	0.336		256	
Crime: Others, Including Drug-Related	0.453		256	

Table 1: Summary Statistics

Source: Authors' tabulations from the NCSC/ICPSR Hung Juries Dataset (Hannaford-Agor et al., 2003). "Crime: Robbert, Burglary, Arson" category also includes larceny, theft, and assault. "Crime: Others, including drug-related" includes child neglect, illegal drug-posession/sale, DUI/DWI, weapons possession, forgery, and unlawful fight.

wide variation within many trials as well. Slightly more than half of the seated jurors in our sample are non-white. This ranges from 84% in the Bronx to 20% in Maricopa. These differences across counties in the racial composition of seated juries largely correspond with the different racial compositions of the populations of these counties. The non-white category is primarily composed of blacks in D.C. and Hispanics in Maricopa, with both groups accounting for a large share in the Bronx and Los Angeles.

For nearly two thirds of the trials in our sample, attorneys enjoyed high participation in voir dire. This includes all of the trials in the Bronx, compared to 27% of trials in Los Angeles and about 60% of trials in D.C. and Maricopa. Supplemental Juror Questionnaires were available much less frequently (20% of trials) and were never available in Maricopa. Just over two thirds of the trials in our sample granted either privilege to attorneys, which is our main indicator of attorney empowerment.

Judges tended to give defense attorneys slightly higher skill ratings than prosecutors on average,

while the average of juror ratings for the prosecutor for a given trial exceeded that for the defense attorney on average across the trials in our sample. Our main judge plus juror indicator of relative attorney skill suggests that the defense attorney had a relative skill advantage over the prosecutor in 46% of trials.³²

About 15% of trials our sample involve a defendant charged with murder or manslaughter and about a third of the trials involve property-related crimes like robbery and arson. The representation of sexual crimes is small in our sample, and drug-related and other crimes such as weapons or forgery charges constitute slightly less than half of our trials.

4 Empirical Results

As has been discussed in detail, our theoretical model yields two predictions. The first is straightforward. When attorneys are empowered and the defense attorney is relatively more skilled, seated jurors will be more likely to be predisposed towards the defense. We first present and discuss results that confirm this prediction for the trials in our dataset. We interpret these results as confirmation of the basic validity of our model.

The second prediction is less amenable to empirical verification. The model suggests that skilled and empowered defense attorneys can end up manipulating the racial composition of juries in various ways in their pursuit of jurors with inclinations towards their side. The precise direction and magnitude of the effect on racial composition depends upon the mechanisms of empowerment that are operative and on the information that the attorneys for a given trial are able to learn about potential jurors, which we cannot observe. Therefore, if we observe skill and empowerment affecting the racial composition of juries in a certain direction, we cannot view this as confirmation of this prediction. Our approach will be to first test for a racial composition effect, and then to infer from this what we can about the benefits that attorneys are able to obtain from empowerment, under the assumption that our theoretical model is correct. We also provide supporting evidence for our inferences by examining evidence on the number of strikes exercised during voir dire. Finally, we provide evidence that the empowerment's effects on the predispositions of the seated jurors are reflected its impact on trial verdicts.

4.1 Supporting Evidence for the Exogeneity of Empowerment

Before testing the effect of empowerment on juror bias and race, we provide some supporting evidence for the exogeneity of empowerment in relation to key characteristics of the trial. Table 2 reports estimates

³²This is perhaps unexpectedly low, as there is a general perception of resource-constrained District Attorneys' Offices being at a disadvantage relative to large private law firms representing defendants. Indeed, the defense attorneys in our dataset tend to be older and more experienced than their counterparts. However, over half of the trials in our sample involved a public defender (ranging from a third of trials in the Bronx to nearly three quarters in Los Angeles), and there is also a general perception of Public Defenders' Offices competing ineffectively against private law firms for talented defense attorneys. In addition, anecdotal evidence suggests that exceptional law students are attracted to District Attorneys' Offices out of law school, because of the opportunity to gain trial experience early in their careers.

from a linear probability model predicting attorney empowerment in voir dire.³³ Attorney empowerment within a trial is not correlated with any case or attorney characteristics with the exception of the public defender indicator and county fixed effects. In all of our specifications, we control for the presence of a public counsel for the defense and county dummies.

Most importantly, empowerment is associated neither with the judge's assessment of evidence in the trial nor his/her assessment of attorney skill. We interpret results in Table 2 as providing credible support for our identification assumption that empowerment is exogenous to broad aspects of the trial and the judge's assessment of the case and attorneys.

4.2 Attorney Empowerment and Jury Predispositions

When attorneys are empowered in voir dire, are jurors more likely to be favorably disposed to the side with the more skillful attorney? In order to analyze this question, we estimate the following equation

Predisposition_{*jtc*} =
$$\beta_0$$
 + β_1 Empowered + β_2 Def. More Skilled + β_3 {Empowered × Def. More Skilled}
+ $\beta_4 X_j + \beta_5 \Pi_t + \beta_6 \Omega_c + \varepsilon_{jtc}$, (29)

for juror j in trial t in county c, controlling for individual juror characteristics X, trial-level characteristics Π , and county-fixed effects Ω . Each observation is an individual juror, and standard errors are adjusted to account for intra-trial correlation. Our chosen indicator of juror predisposition is measured on an ordinal seven-point scale. We report results from an ordinary least squares specification that provides a more straightforward interpretation of marginal effects as well as an ordered logistic specification that accounts for the ordinal nature of our dependent variable. The main explanatory variables of interest are our measure of attorney empowerment, our measure of relative attorney skill, and most importantly, their interaction.

Table 3 presents our main results using the judge and juror hybrid indicator of relative attorney skill. The first three columns present OLS results while the last three columns present the ordered logistic results. The first and fourth columns of Table 3 purposely omit the interaction between attorney empowerment and relative skill. The results suggest that attorney empowerment and the presence of a more skilled defense attorney are separately associated with large and statistically significant increases in the probability that a seated juror will be predisposed towards the defense.³⁴ Taking these results at face value for the moment, this is an unexpected result when considering the commonly notion in the legal community. Legal scholars and commentators typically assume that greater attorney involvement in voir dire is detrimental to the defense.

The second and fifth columns of Table 3 report what the theoretical model suggests is the "correct"

³³Results from probit and logistic regression are the same. The decreased sample size is due to the low attorney survey response rates.

 $^{^{34}}$ Columns (4) through (6) in the table presents the ordered logistic estimated coefficients. These are marginal effects on the cumulative log-odds ratio of a juror's response being higher on the seven-point scale than a given cut-off category.

Dep. Var.= Empowered (1 if either SJQ or High Attorney Involvement)								
Judge Assessment of Evidence and Attorney Skill								
Judge: Evidence favored $(1 = Pro. to 7 = Def.)$	-0.037							
	(0.027)							
Judge: Attorney Skill Difference	-0.006							
	(0.029)							
Case Characteristics								
Black Defendant	0.103							
	(0.091)							
Victim Present	-0.193							
	(0.147)							
Female Victim	0.105							
	(0.124)							
Victim Not Black	-0.112							
	(0.125)							
Crime: Murder and Manslaughter	-0.080							
	(0.193)							
Crime: Robbery, Burglary, Arson	-0.065							
	(0.170)							
Crime: Others, including drug-related	-0.223							
	(0.197)							
Attorney Characteristics								
Public Defender	0.220**							
I ublic Delender	(0.230)							
Vears Practice (Defense)	0.008							
rears r ractice (Derense)	(0.003)							
Years Practice (Prosecutor)	-0.005							
	(0.010)							
Age (Defense)	0.001							
	(0.006)							
Age (Prosecutor)	-0.002							
<u> </u>	(0.008)							
Female (Defense)	0.094							
	(0.092)							
Female (Prosecutor)	0.042							
	(0.082)							
County								
	0 691***							
LA	$-0.021^{-0.021}$							
Mariaana	(0.130)							
мансора	-0.022							
DC	(0.101)							
	-0.409							
	(0.113)							
NT	1.40							
IN D ²	142							
n	0.301							

Table 2: Determinants of Empowerment (OLS Regression)

Notes: Robust standard errors in parentheses. p < 0.10, **p < 0.05, ***p < 0.01. The excluded county category is Bronx and the excluded crime category is sexual crimes. "Crime: Robbert, Burglary, Arson" category also includes larceny, theft, and assault.

Dep. Var. = Juror Predisposition $(1=P \text{ to } 7=D)$									
	OLS			Ordered Logistic					
	(1)	(2)	(3)	(4)	(5)	(6)			
Empowered	0.550^{***}	0.337	0.250	0.461^{***}	0.264	0.197			
	(0.167)	(0.207)	(0.183)	(0.155)	(0.194)	(0.172)			
Empowered \times Def. More Skilled		0.488^{*}	0.700^{***}		0.458^{*}	0.622^{***}			
		(0.292)	(0.186)		(0.269)	(0.169)			
Def. More Skilled	0.548^{***}	0.216		0.475^{***}	0.168				
	(0.150)	(0.233)		(0.137)	(0.216)				
Judge: Evidence Favored $(1=P \text{ to } 7=D)$	0.297^{***}	0.299^{***}	0.306^{***}	0.268^{***}	0.271^{***}	0.277^{***}			
	(0.055)	(0.055)	(0.054)	(0.052)	(0.053)	(0.052)			
Juror Non-White	0.373^{***}	0.373^{***}	0.371^{***}	0.334^{***}	0.330^{***}	0.325^{***}			
	(0.110)	(0.110)	(0.110)	(0.099)	(0.100)	(0.099)			
R^2	0.13	0.13	0.13	n/a	n/a	n/a			
Log-Likelihood	-3908	-3905	-3906	-3449	-3446	-3446			
# Jurors	1883	1883	1883	1883	1883	1883			
# Trials	256	256	256	256	256	256			

Table 3: Attorney Empowerment and Jury Predisposition Using Judge + Juror Assessment of Skill

Notes: Robust standard errors, clustered at the trial level, are in parentheses. All trials with non-missing data for at least one juror are included. Each equation includes controls for county, crime type, and defendant and victim characteristics, all of which have been suppressed in the table, along with the estimated ordered logistic cut points. The excluded relative attorney skill category is "prosecutor more skilled." *p < 0.10, **p < 0.05, ***p < 0.01.

Table 4:	Attorney	[·] Empowerment	and Jury	v Predispo	sition Using	Judge	Assessment	of	Skill
T (0) 1 (1)	110001110,	Diffoonormonio	and our	, rroumpo	DIGIOII COMIN	ouugo.	LODODDIIIOIIO	U 1	~ IIIII

Don Von - Lunor Predignosition (1-D to 7-D)										
Dep. val. – Juloi riedisposition (1=r to 7=D)										
		OLS		Or	stic					
	(1)	(2)	(3)	(4)	(5)	(6)				
Empowered	0.564^{***}	0.170	0.303	0.470***	0.089	0.221				
-	(0.170)	(0.245)	(0.220)	(0.156)	(0.220)	(0.201)				
Empowered \times Def. More Skilled		0.758^{**}	0.433**		0.730**	0.397^{**}				
		(0.324)	(0.211)		(0.293)	(0.190)				
Def. More skilled	0.190	-0.327		0.151	-0.334					
	(0.174)	(0.261)		(0.155)	(0.235)					
Judge: Evidence Favored (1=P to 7=D)	0.319^{***}	0.315^{***}	0.308^{***}	0.288^{***}	0.287^{***}	0.278^{***}				
	(0.056)	(0.056)	(0.055)	(0.053)	(0.054)	(0.052)				
Juror Non-White	0.371^{***}	0.370^{***}	0.370^{***}	0.329^{***}	0.328^{***}	0.329^{***}				
	(0.110)	(0.108)	(0.109)	(0.098)	(0.097)	(0.097)				
R^2	0.12	0.12	0.12	n/a	n/a	n/a				
Log-Likelihood	-3919	-3912	-3914	-3458	-3451	-3453				
# Jurors	1883	1883	1883	1883	1883	1883				
# Trials	256	256	256	256	256	256				

Notes: Robust standard errors, clustered at the trial level, are in parentheses. All trials with non-missing data for at least one juror are included. Each equation includes controls for county, crime type, and defendant and victim characteristics, all of which have been suppressed in the table, along with the estimated ordered logistic cut points. The excluded relative attorney skill category is "no skill difference." *p < 0.10, **p < 0.05, ***p < 0.01.

specification, and the results broadly confirm basic predictions from our theoretical model.³⁵ In column (2), the estimated coefficient on the interaction term is 0.49, implying that, when the defense attorney is more skilled, attorney empowerment leads to an increase of about a half a point on the seven-point favoritism scale (or about 0.25 standard deviations) for the average seated juror.³⁶ Empowerment's effect on the greater levels of jurors' favorability for the defense is predominantly accounted for by the trials in which the defense is more skilled. These results suggest that when attorneys are empowered *and* the defense attorney is relatively more skilled, there is a significant and substantially higher probability that seated jurors will be favorably inclined towards the defense. The combination of skill and empowerment is crucial to deliver this effect in the model: empowerment can provide both attorneys with a greater ability to strike undesirable potential jurors, but it is the combination of this with the greater skill of the defense attorney that sharpens the defense attorney's striking ability and allows it to be directed more profitably.

The model also predicts, when taken very literally, the defense attorney's greater skill should have no effect on the probability that a seated juror has favoritism for the defense when attorneys are not empowered. This is the case because the model was constructed such that skill is only useful to attorneys in utilizing the benefits that come with empowerment. The results in columns (2) and (5) of Table 3 broadly confirm this aspect of the model in substance: the estimated coefficient on "Defense More Skilled" is positive but statistically insignificant. On the other hand, it would also seem reasonable that the skill of defense attorneys could be partially manifested in influencing juror inclinations through arguments during trial, which the imprecisely-estimated positive coefficient could be capturing.

The reason that the results in the second and fifth columns are only a qualified confirmation of the model lies in the construction of our measure of relative attorney skill. The excluded category of this dummy variable represents trials in which the prosecutor was strictly more skilled than the defense attorney. The theoretical model is reversible according to which attorney is assumed to have the greater skill, so the opposite prediction should hold when it is the prosecutor who is more skilled: the probability that a seated juror is predisposed towards the *defense* should be *lower* when attorneys are empowered and the prosecutor is more skilled. Therefore, one would have expected the estimated coefficient on the main empowerment effect to be negative. Instead, the results in the second column show that this effect is positive, though not statistically different from zero. This suggests that prosecutors in our dataset are less able to leverage empowerment and a skill advantage in jury selection than their colleagues across the aisle, though potential explanations for this remain elusive.

The third and sixth columns of Table 3 take the theoretical model literally and impose a zero effect of attorney skill when attorneys are not empowered. The effect of empowerment when the defense attorney is the more skilled becomes larger and more statistically significant, while the effect when the

³⁵The estimated ordered logistic cut-points for the specifications in Table 3 do not indicate a great deal of non-linearity in the underlying latent favoritism variable, lending some justification to treating the raw seven-point variable as a continuous, cardinal measure.

³⁶Results are also similar for random effects specifications, either GLS with the dependent variable in its original form, or probit, with the dependent variable transformed to a dummy variable for favoritism ratings of 5 or higher versus 4 or lower.

prosecutor is the more skilled remains positive but is closer to zero.

There are some important effects associated with the control variables that should be briefly examined. These effects are nearly identical across the three specifications. First, non-white jurors have a significantly and substantially higher likelihood of being predisposed to the defense than white jurors. This corresponds with a primitive assumption of our theoretical model and with much anecdotal and empirical evidence. Second, when the evidence is strong for the defense's case in the judge's opinion, seated jurors are much less likely to report having favored the defense, as expected. Perhaps the most interesting aspect of the estimated effects of strength of evidence is the context they provide for interpreting the magnitude of the effect of the interaction between empowerment and skill.³⁷ The specification in the third column suggests that the successes of skilled and empowered defense attorneys in jury selection can be substantial enough to offset the disadvantage from being faced with evidence supporting the prosecution. Point estimates suggest that empowerment and defense skill compensate for about a one standard deviation increase in the judge's evaluation of evidence in favor of the defense. This is a potentially unsettling finding from the perspective of debates on the role of jury selection in the service of justice.³⁸

Table 4 repeats analysis in Table 3 using only the judge's assessment attorney skill to determining the relative skillfulness of attorneys within a trial. The excluded skill category in relative attorney skill is now "no skill difference," and we separately control for "prosecutor more skilled" and its interaction with empowerment. Estimates strongly confirm our findings in the previous table. While the standard errors are slightly larger for our main coefficients of interest, the coefficient on the interaction between empowerment and defense more skilled indicates a larger impact of empowerment on juror bias when the defense attorney is more skillful. Coefficients on the interaction between empowerment and "prosecutor more skilled" is negative but statistically insignificant at conventional levels, again indicating that the defense is able to better leverage empowerment during the jury selection process.

4.2.1 Distribution of Juror Predisposition by Empowerment and Relative Attorney Skill

Figure 1 and Table 5 assess whether there are any distributional impacts of empowerment on juror bias and provide strong support for our conclusions in Tables 3 and 4. Figure 1 plots the kernel density of residuals from the an OLS regression predicting juror predisposition controlling for all the controls in Table 4 except for empowerment, attorney skill, and their interactions. Results strongly confirm our

³⁷In specifications where we separately identify those trials in which the judge's assessment of the evidence favored the prosecution versus the defense, we find that when the evidence is strong for the prosecution's case, seated jurors are much less likely to report having favored the defense. On the other hand, when the evidence is strong for the defense's case, seated jurors are only somewhat more likely to report a predisposition towards the defense, and this effect is not statistically significant. This result is sensible in direction, and though somewhat weak, should be considered in the context of the purposeful intention of the judicial system to avoid false convictions, which could lead many jurors to view weak or offsetting evidence (the excluded category) as being in favor of the defendant.

³⁸Controls for county, type of crime, and defendant and victim characteristics have been suppressed in the table, and for the most part are not associated with important effects. One interesting result is that, conditional on there being a victim of the alleged crime, if the victim was female, jurors were significantly less likely to report a predisposition towards the defense.



Figure 1: Residual Juror Bias by Attorney Skill Difference and Empowerment

Notes: Kernel(epanechnikov), bandwidth(0.75). Attorney skill difference is determined using the judge's assessment of skills only. Residual bias is the difference between juror predisposition and predicted juror predisposition from a linear regression controlling for judge opinion, county, crime type, defendant race, whether defense attorney is a public defender, juror characteristics, and victim characteristics.

Dep. Var. $=$ Residual Juror Bias										
			Percentiles							
	Mean	St.Dev.	10th	25th	50th	75th	90th			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)			
Unempowered: No Skill Difference	-0.044	1.735	-2.186	-1.476	-0.129	1.175	2.442			
Empowered: No Skill Difference	0.036	2.045	-2.514	-1.708	-0.132	1.559	3.090			
Unempowered: Def. More Skilled	-0.342	1.888	-2.600	-1.936	-0.601	1.167	2.518			
Empowered: Def. More Skilled	0.447	2.028	-2.358	-1.213	0.436	2.081	3.136			

Table 5: Distribution of Residual Juror Bias

Notes: Residual bias is the difference between juror predisposition and predicted juror predisposition from a linear regression controlling for judge opinion, county, crime type, defendant race, whether defense attorney is a public defender, juror characteristics, and victim characteristics. Attorney skill difference is determined using the judge's assessment of skills only.

findings in Tables 3 and 4. When there is no difference in the attorneys' skill (as assessed by the judge), there is a zero impact of empowerment on the mean or the entire distribution of juror bias. However, in trials in which defense is more skilled, empowerment is associated with a strong rightward shift in the distribution of the residual juror bias towards the defense. There is no difference in the overall variation of juror bias as measured by the standard deviation, however, the differences in residual juror bias is largely concentrated in the higher percentiles of juror bias favoring defense.

4.3 Evaluating our Measure of Attorney Skill

As discussed in the previous section, our primary measure of attorney skill is based on an index of relative skill constructed from survey questions asked of judges and jurors. Table 6 shows the results of linear regressions of the hybrid skill index and the judge's rating on the few attorney characteristics for which data are available.³⁹ Previous legal experience of the defense attorney is associated with a greater relative skill rating. This effect is statistically significant but small. Previous legal experience of the prosecuting attorney has a somewhat offsetting negative effect, as would be expected, but this is small and not statistically significant. In addition, specific criminal experience for either attorney is not associated with an important effect on relative skill.

On the other hand, there is a large positive and sometimes significant effect on the skill index when the defense attorney is a woman. We interpret this as potentially embodying a selection effect: female defense attorneys entrusted with lead counsel duties are likely to be more highly skilled and talented than their male counterparts if they suffer statistical discrimination throughout their career.⁴⁰ Thus, our chosen measure of relative attorney skill appears to be a valid indicator of at least some objective aspects of attorney skill.⁴¹

The results in Table 6 suggest a natural approach to diagnosing and correcting for any simultaneity between juror predispositions and our measure of attorney skill. Attorney experience and gender are correlated with the skill ratings of attorneys, but should be exogenous to the way that jurors responded to their surveys, making these variables good candidates to create a predicted attorney skill index. However, we encounter two problems when attempting to implement such a strategy. First, the sample size is smaller because the number of trials for which attorney characteristics are available is significantly lower than for other variables. Second, these attorney characteristics are not strong proxies for the attorney skill measure. Both of these problems lead to imprecisely-estimated coefficients, to such a degree that statistical comparisons with our main results on predispositions are not meaningful.

³⁹Each observation in the regressions in Table 6 is a trial, because all variables vary only at the trial level. As mentioned previously, attorney response rates were low compared to other parties, so the number of trials represented in this table is lower than in others.

⁴⁰See Lehmann (2011) for a discussion of this effect for blacks in large law firms.

⁴¹The same effects appearing in Table 6 also appear in specifications focusing on the separate ratings for the defense or prosecution by judges and jurors. In addition, some specifications examining skill ratings reported by jurors indicate that higher juror religiousness is associated with a greater likelihood of giving the prosecutor a high skill rating. Other juror characteristics – most notably race – are not significantly correlated with the skill ratings given to either attorney. Since juror predispositions are correlated with race but not religiousness in our data, this suggests that the skill ratings given by jurors are not necessarily biased by their predispositions.

Dep. Var. = Difference in Attorney Skill (Defense minus Prosecutor Rating)									
	Judge +	Jurors	Juo	dge					
	(1)	(2)	(3)	(4)					
Years Practice (Defense)	0.057***	0.035^{*}	0.070***	0.068***					
	(0.016)	(0.021)	(0.017)	(0.021)					
Years Practice (Prosecution)	-0.016	-0.023	-0.015	-0.021					
	(0.028)	(0.033)	(0.029)	(0.033)					
Previous Criminal Trials (Defense)	-0.001	-0.001	-0.000	-0.000					
	(0.001)	(0.001)	(0.001)	(0.001)					
Previous Criminal Trials (Prosecution)	0.001	0.002	0.002	0.002					
	(0.003)	(0.003)	(0.003)	(0.003)					
Age (Defense)	-0.006	-0.003	-0.011	-0.015					
	(0.012)	(0.013)	(0.013)	(0.015)					
Age (Prosecution)	0.003	0.005	-0.011	-0.011					
	(0.020)	(0.022)	(0.021)	(0.022)					
Female (Defense)	0.373	0.408	0.395^{*}	0.292					
	(0.227)	(0.273)	(0.238)	(0.283)					
Female (Prosecutor)	0.025	0.053	0.123	0.104					
	(0.215)	(0.231)	(0.227)	(0.232)					
Controls	No	Yes	No	Yes					
\mathbf{p}^2	0.00	0.14	0.14	0.00					
<i>K</i> ²	0.08	0.14	0.14	0.20					
# Irials	160	147	158	145					

Table 6: Determinants of Attorney Skill Difference (OLS Regressions)

Notes: Robust standard errors are in parentheses. Each equation includes a constant term, which has been suppressed in the table. In addition, the second and third columns include controls for county, crime type, and selected defendant and victim characteristics, which have also been suppressed in the table. See the text for a description of the dependent variable. *p < 0.10, **p < 0.05, ***p < 0.01.

Finally, we have employed measures of absolute attorney skill in some specifications in order to broaden our focus beyond simple skill asymmetries. The results suggest that empowerment increases the likelihood that a seated juror will be predisposed to the defense to a greater degree when the defense attorney has a higher skill rating from the judge. The effects of the prosecutor's skill rating are very imprecisely estimated, and more generally, these specifications have lower explanatory power than our main specifications. This is perhaps an indication that the attorney skill ratings, measured on a subjective seven-point scale, are more applicable for making relative comparisons within a trial.

4.3.1 Summary

We have discussed evidence that the more highly skilled attorney can indeed leverage empowerment in voir dire to retain jurors based on their predispositions. This result matches the main prediction of our theoretical model. Our results have been shown to be robust to different specifications and alternative measures of attorney empowerment. We now proceed to examine further predictions of our model,

Dep. Var. $= 1$ if Non-White, 0 if White										
	Judge +	Jurors Ratir	ng of Skill	Judg	Judge Rating of Skill					
	(1)	(2)	(3)	(4)	(5)	(6)				
Empowered	0.027	0.026	0.036	0.029	0.031	0.009				
	(0.027)	(0.035)	(0.030)	(0.026)	(0.035)	(0.031)				
Empowered \times Defense More Skilled	. ,	0.002	-0.021		0.022	0.021				
		(0.045)	(0.025)		(0.050)	(0.029)				
Defense More Skilled	-0.022	-0.023		0.015	-0.000	· · ·				
	(0.021)	(0.038)		(0.023)	(0.040)					
LA	-0.157***	-0.157***	-0.157***	-0.152***	-0.150***	-0.156***				
	(0.037)	(0.037)	(0.037)	(0.037)	(0.037)	(0.037)				
Maricopa	-0.569***	-0.569***	-0.570***	-0.569***	-0.566***	-0.567***				
	(0.037)	(0.037)	(0.037)	(0.037)	(0.037)	(0.037)				
DC	-0.238***	-0.238***	-0.236***	-0.238***	-0.239***	-0.234***				
	(0.034)	(0.034)	(0.035)	(0.035)	(0.035)	(0.035)				
R^2	0.27	0.27	0.27	0.27	0.27	0.27				
# Jurors	1883	1883	1883	1883	1883	1883				
# Trials	256	256	256	256	256	256				

Table 7: Prediction 2: Empowerment and Jury Race (OLS Regressions)

Notes: Robust standard errors, clustered at the trial level, are in parentheses. Each specification includes a constant term and controls for non-race juror characteristics, crime type and defendant and victim characteristics, all of which have been suppressed in the table. The excluded relative attorney skill category for columns (1) through (3) is "prosecutor more skilled' and for columns (3) through (6), "no skill difference." *p < 0.10, **p < 0.05, ***p < 0.01.

retaining our preferred measures of attorney empowerment and skill.

4.4 Attorney Empowerment and Jury Race

In this section, we ask whether attorney empowerment has an impact on the racial composition of the seated jury. Table 7 examines the effect of attorney empowerment and skill on the probability that a seated juror is non-white.⁴² Our treatment of the two is the same as in Table 3. Column (1) includes the main effects only; column (2) introduces their interaction in addition; and column (3) retains the interaction effect but restricts the main skill effect to be zero. The dependent variable is a dummy variable, with whites as the excluded category. All specifications include several trial-level controls.⁴³ Each observation is an individual juror, and standard errors are adjusted to account for intra-trial correlation. We report OLS results, but results from non-linear specifications are similar.⁴⁴

The main message in Table 7 is that there is no effect of attorney empowerment in voir dire on the racial composition of seated juries, regardless of attorney skill. The estimated coefficients for the main

⁴²Results from a logit specification are the same and are reported in the Appendix Table A.2.

⁴³Juror characteristics have been omitted from these regressions to avoid simultaneity with the dependent variable.

⁴⁴This functional form assumption does not appear to be important. Results are similar for linear specifications and for logit and probit specifications with or without random effects. Further, multinomial logistic results do not indicate important differences for Blacks and Hispanics, the two primary non-white racial groups in the data.

effects and the interaction term are small in magnitude and statistically insignificant. Instead, the county fixed effects account for much of the explanatory power of the regressions. That is, although attorneys appear to have an impact on the average bias of the jurors they retain when they are empowered, the racial composition of the seated jury is driven almost entirely by the demographics of the jury pool and other county-level effects.⁴⁵

The marginal effect of attorney empowerment on the probability of a seated juror being non-white when the defense is more skilled from the specification in column (2) is 0.2 percentage points, with a 95% confidence interval of -8.6 percentage points to 9.1 percentage points. On a twelve-person jury, this range amounts to the exclusion or inclusion one additional non-white juror, at most. In contrast, moving a trial from D.C. County (with a 51% non-white population) to Bronx County (with a 84% non-white population) would increase the estimated probability of a seated juror being non-white by 24 percentage points on average, or 2.9 non-white jurors on a twelve-person jury.⁴⁶

Our theoretical model does not predict a zero effect on racial composition of seated juries. Rather, it predicts an effect – potentially large – in either direction depending on how attorneys leverage empowerment. If empowerment is effective simply by allowing attorneys to find convincing grounds to have jurors dismissed for cause, a more highly skilled defense attorney will strike more white panelists than otherwise. On the other hand, if empowerment provides attorneys with strong information regarding juror leanings (in addition to increasing striking ability or not), a more highly skilled defense attorney could choose to retain only those jurors appearing to be predisposed to the defense, regardless of race, and thus end up with a higher proportion of whites on the seated jury than would otherwise be the case. In the context of these predictions, we interpret our empirical results as indicating a mix of these various empowerment effects across the trials in our sample. That is, in at least some of the trials, attorneys were able to leverage empowerment to learn valuable information about jurors that would otherwise remain unobserved.

5 Additional Evidence for the Effect of Empowerment

5.1 Supporting Evidence from the Number of Panelists Struck

Corollary predictions concerning the number of strikes exercised during voir dire can also be extracted from our model. As with predictions about race, these predictions about number of strikes are dependent upon the specific functions of empowerment. For example, when empowerment only increases attorney striking ability, Assumption 2 implies that more panelists will be struck in expectation, as both attorneys will have a greater striking success rate than otherwise. If, instead, empowerment only

⁴⁵These conclusions remain true when considering the alternative measures of attorney empowerment explored in the previous section.

⁴⁶The change in racial composition due to empowerment, though small in a practical sense, may nonetheless be important for trial outcomes. Our findings in Lehmann and Smith (2013), however, do not suggest that this is the case. As reported there, we find that changes in the proportion of blacks on seated juries have a smaller effect on and explain less of the variation in convictions and acquittals per trial than measures of evidentiary strength and complexity.

Dep. Var. $=$ Total Number of Strikes											
	Judge +	- Jurors Rating	g of Skill	Juc	Skill						
	(1)	(2)	(3)	(4)	(5)	(6)					
Empowered	1.4302	3.2589	2.6646^{*}	1.5619	4.3501^{*}	1.3134					
	(1.4305)	(1.9767)	(1.5937)	(1.4322)	(2.2616)	(1.7137)					
Empowered \times Def. More Skilled		-4.2067	-2.8108		-6.2510^{*}	-1.7135					
		(3.1390)	(2.0204)		(3.6543)	(2.4732)					
Def. More Skilled	-1.5314	1.3925		0.1167	4.5018						
	(1.6089)	(2.4325)		(1.9357)	(2.7321)						
Panel Size	0.4060^{***}	0.4108^{***}	0.4086^{***}	0.4094^{***}	0.4136^{***}	0.4078^{***}					
	(0.1343)	(0.1333)	(0.1331)	(0.1330)	(0.1313)	(0.1320)					
LA	-11.1815^{***}	-11.1185^{***}	-11.1625^{***}	-10.8354^{***}	-10.6320^{***}	-11.2476^{***}					
	(3.5499)	(3.5470)	(3.5524)	(3.4743)	(3.4664)	(3.5420)					
Maricopa	-7.4223^{*}	-7.5598*	-7.5348*	-7.2436^{*}	-7.7642^{**}	-7.5651*					
	(4.0054)	(4.0118)	(4.0112)	(3.9293)	(3.9063)	(3.9629)					
DC	1.6913	1.7789	1.6828	1.5499	1.4650	1.5774					
	(2.8830)	(2.8662)	(2.8830)	(2.8962)	(2.8836)	(2.9064)					
R^2	0.49	0.50	0.50	0.50	0.50	0.50					
# Trials	237	237	237	237	237	237					

Table 8: Number of Strikes (OLS Regressions)

Notes: Robust standard errors are in parentheses. All trials with non-missing data for at least one juror are included. Each equation includes a constant term and controls for crime type, defendant and victim characteristics, and length of voir dire, all of which have been suppressed in the table. The excluded relative attorney skill category for columns (1) through (3) is "prosecutor more skilled' and for columns (3) through (6), "no skill difference." *p < 0.10, **p < 0.05, ***p < 0.01.

provides information on juror predispositions and the defense attorney chooses to strike white panelists exhibiting the low signal only, there will be fewer panelists struck in expectation, as no white panelists exhibiting the high signal will be struck.

Table 8 examines the effect of attorney empowerment and skill on the number of strikes exercised during voir dire. The same specifications as in Table 7 examining race effects are employed, with the inclusion of additional controls for other aspects of the voir dire process. The dependent variable is the total number of strikes exercised, including all strikes for cause and the number of peremptory strikes exercised by each side.⁴⁷ Each observation is a trial, and we report OLS estimates.⁴⁸

The results in columns (2) and (5) suggest that when empowered in voir dire, defense attorneys strike fewer panelists when they have a skill advantage compared to prosecutors. Large standard errors prevent us to establish statistical significance at conventional levels when using the hybrid skill rating, but using the judge's assessment of attorney skill, the effect of empowerment when defense is more

⁴⁷Strikes for cause outnumber total peremptory strikes for many trials, but for some cases, the number of strikes for cause is reported as zero. We suspect that these are cases for which the entire venire was examined for cause out of the presence of court clerks, before the sequential exercise of peremptory challenges was begun (and also that the panel size reported for these trials refers to the size of the remaining venire after strikes for cause). We caution that the data on strikes for cause are therefore likely to be incomplete. However, results are similar when each of these three categories of strikes is examined individually, so we err on the side of using the more comprehensive dependent variable.

⁴⁸Estimates from poisson regressions show similar results.

skillful is significant at the 10-percent level. however, both sets of coefficients are estimated imprecisely, so that fairly substantial effects in either direction cannot be ruled out. Most of the explanatory power of the regressions comes from the county fixed effects. Judge tendencies and state guidelines generally lead to fewer potential peremptory challenges being available to attorneys in Los Angeles and Maricopa compared to the other counties.

In principle, we could attempt to better characterize how the attorneys in our dataset leveraged empowerment in voir dire by examining the model's joint predictions concerning race and number of strikes, and comparing them with our empirical results. For example, if we had found that empowerment was associated both with a lower proportion of non-whites on seated juries and with a lower number of strikes in trials for which the defense attorney had a skill advantage, we could infer that attorneys were often in the situation like the second example mentioned above, i.e. with the information aspect of empowerment dominating, and with the high signal rare enough in the population that the defense attorney attempts to retain all but white panelists exhibiting the low signal.

However, our finding that empowerment is associated with a decrease in number of strikes exercised when the defense more skillful, provides some supporting evidence for the presence of the information effect of empowerment. The small and insignificant effects on the racial composition and a significantly negative effects on the number of strikes, albeit imprecisely measured, could indicate varying prominence of the striking ability and information functions of empowerment across trials, along with variation in the quality of the signals of juror predispositions that skilled and empowered attorneys were able to obtain. In this sense, the results for number of strikes do provide some corroborating evidence in support of the results for racial composition.

5.2 Empowerment and Trial Outcomes

Our focus on attorney empowerment's effects on juror predispositions directly rather than on trial verdicts is motivated by the concern that verdicts are the result of a complex process of group decisionmaking. This black-box surrounding the dynamics of group deliberation implies that techniques available to attorneys for assessing individual jurors tend to be much more advanced and utilized in practice than those predicting interaction effects between jurors in voir dire (Gobert et al. 2009). Nevertheless, despite these caveats, it would be useful to assess whether individual jurors' biases translate into impacts on verdicts, an outcome that policy-makers may ultimately care about the most.

Table 9 assesses whether attorney empowerment and skill differences are associated with variations in individual jurors' voting patterns on the most serious charge in the trial. Our findings show that empowerment's effect on jurors' predispositions are also reflected in the variations in their voting patterns. In columns (1) through (3) relying on the judge and juror hybrid ratings as the determinant of attorneys' relative skills, we find that in trials with a more skillful defense attorney, empowerment is associated a significant increases in the likelihood of a "not guilty" vote on the most serious charge. As in the predisposition results, empowerment does not appear to have any significant impact on voting patterns when the prosecutor is more skilled in the trial. Using the judge's rating, the direction of

Dop. Var 1 if Voted Not Chilty, 0 if Voted Chilty										
Dep. val. –	Judge +	Jurors Rati	ng of Skill	Judg	Skill					
	(1)	(2)	(3)	(4)	(5)	(6)				
Empowered	0.130**	0.048	0.034	0.133**	0.102	0.144**				
Empowered × Defense More Skilled	(0.060)	(0.072) 0.190*	(0.065) 0.225***	(0.060)	(0.085) 0.109	(0.070) 0.053				
Empowered × Detense More Skilled		(0.104)	(0.065)		(0.124)	(0.055) (0.078)				
Defense More Skilled	0.164^{***}	0.036		0.018	-0.056					
Judge: Evidence Favored (1=P to 7=D)	(0.055) 0.094^{***}	(0.084) 0.095^{***}	0.096***	(0.003) 0.103^{***}	(0.098) 0.103^{***}	0.104***				
Prosecutor More Skilled	(0.020)	(0.020)	(0.020)	(0.020) -0.096	(0.020) -0.091	(0.020)				
Empowered × Prosecutor More Skilled				(0.068)	(0.102)	-0.103				
Empowered × 1 rosecutor more Skilled					(0.134)	(0.088)				
R^2	0.22	0.23	0.23	0.20	0.21	0.20				
# Jurors	1883	1883	1883	1883	1883	1883				
# Trials	254	254	254	254	254	254				

Table 9: Juror Voted Not Guilty on Most Serious Charge (OLS Regressions)

Notes: Robust standard errors, clustered at the trial level, are in parentheses. Each specification includes a constant term and controls for crime type and defendant and victim characteristics, and county fixed effects all of which have been suppressed in the table. The excluded relative attorney skill category for columns (1) through (3) is "prosecutor more skilled' and for columns (3) through (6), "no skill difference." *p < 0.10, **p < 0.05, ***p < 0.01.

Don $Var = 1$ Acquitted on Hung 0 if Not										
Dep. val. – 1 Acquitted of Hung, 0 in Not										
	Judge +	Jurors Rati	ng of Skill	Judg	Judge Rating of Skill					
	(1)	(2)	(3)	(4)	(5)	(6)				
Empowered	0.162^{**}	0.057	0.020	0.168^{**}	0.117	0.164^{**}				
	(0.072)	(0.094)	(0.082)	(0.072)	(0.108)	(0.080)				
Empowered \times Defense More Skilled	· · · ·	0.219^{*}	0.302***	· · · ·	0.242	0.157**				
		(0.129)	(0.072)		(0.149)	(0.077)				
Defense More Skilled	0.237^{***}	0.085	· /	0.086	-0.085	· · ·				
	(0.062)	(0.110)		(0.068)	(0.127)					
Empowered \times Prosecutor More Skilled	· · · ·	· /		· · · ·	-0.077	-0.157				
					(0.166)	(0.099)				
Prosecutor More Skilled				-0.125	-0.079	· /				
				(0.080)	(0.134)					
Judge: Evidence Favored (1=P to 7=D)	0.059^{***}	0.061^{***}	0.064^{***}	0.070***	0.071***	0.070***				
	(0.021)	(0.022)	(0.021)	(0.021)	(0.021)	(0.021)				
R^2	0.19	0.20	0.20	0.16	0.18	0.17				
# Trials	254	254	254	254	254	254				

Notes: Robust standard errors, clustered at the trial level, are in parentheses. Each specification includes a constant term and controls for crime type and defendant and victim characteristics, and county fixed effects all of which have been suppressed in the table. Trials in which at least one juror voted to acquit the defendant on the most serious charge is defined as having the most serious charge acquitted. The excluded relative attorney skill category for columns (1) through (3) is "prosecutor more skilled' and for columns (3) through (6), "no skill difference." *p < 0.10, **p < 0.05, ***p < 0.01.

these effects are similar although our estimates are much less precise. In Table 10, we collapse these individual jurors' votes into a case-level verdict on the most serious charge, and we find that empowerment is indeed associated with a 20-percentage point or close to a half standard deviation increase in the likelihood of an acquittal when the defense attorney is more skilled.⁴⁹ The signs and magnitudes of the effects are similar across specifications using either type of attorney skill ratings. Results from these two tables demonstrate that empowerment's effect on pre-deliberations biases had a significant and economically sizable impact on trial outcomes.

6 Alternative Interpretations

We interpret our results as providing reasonable confirmation of our model and of the general hypothesis that attorney empowerment in voir dire can increase the ability of attorneys both to have potential jurors removed and to learn more about them. However, there are alternative explanations for why we might see the interaction of empowerment and skill leading to an effect on juror favoritism but not on juror race.

One such explanation is that attorneys may use voir dire less for striking and learning about jurors and more for testing and laying the groundwork for arguments to be raised during trial. Skilled attorneys afforded this opportunity could potentially be indifferent about the race and any other characteristics of the jurors hearing the case, so long as those jurors could be sufficiently groomed and prepared beginning in voir dire. If attorneys often made use of empowerment in voir dire to this effect, one might expect to see fewer strikes for trials in which attorneys were empowered. However, as discussed above, there are also cases for which our model predicts that the combination of skill and empowerment will be associated with fewer strikes than otherwise. It is, therefore, not clear that these alternative hypotheses about the role of empowerment could be convincingly tested against one another based on number of strikes alone.

Another concern is that our measure of attorney empowerment could be capturing unobserved judge effects. For example, judges who routinely grant in-depth attorney participation in voir dire may hold subtle sympathies for defendants and defense attorneys in general, which could also be reflected in other relevant aspects of the trial, such as the instructions given to the jury, the evidence admitted, and the attorney motions allowed. Unfortunately, the data include neither judge identifiers nor even basic demographic information. Although we find some assurance in the finding that attorney empowerment appears to be uncorrelated with broad aspects of the trials, attorney characteristics, and most importantly, the judge's evaluation of evidence and attorney skill as reported in Table 2, we plan to further address this question in the future by leveraging cross-county variation in the data to test

⁴⁹The charge is assumed to have been acquitted if any juror in the trial voted not guilty on the charge. We rely on this definition, because one, not all jurors in the trials answered the voting question, and two, we do not have an indicator for the verdict on the most serious charge at the case-level. Rather, the survey only provides information on the total number of charges in the case that ended in acquittals or convictions, and not a unique indicator for the most serious charge. However, since only 10% of charges in our estimating sample ended in a hung jury, our measure should be a close proxy to the actual number of acquittals.

selected comparative statics of our model. If the empirical results of this exercise correspond more with the political leanings of judges in certain counties than with the model's predictions, such judge effects may deserve more serious consideration.

Finally, it should be mentioned that few criminal charges end up going to trial and terminating with a jury verdict.⁵⁰ There are many factors at play determining which cases see the courtroom, from the presence of mandatory minimum sentences to chance interactions between attorneys attempting to work out a plea bargain. It is unclear how we might meaningfully assess which of many potential selection effects might be affecting our results, and to what degree. We leave a more detailed investigation of this matter for future work.

7 Conclusion

This paper is the most complete attempt to date to measure what attorneys can actually accomplish, given varying degrees of empowerment in jury selection, in terms of multiple characteristics of the seated jurors they manage to retain. We have constructed a model of attorney behavior in which empowerment can operate through multiple channels, and in which the benefits of empowerment can depend on attorney skill. We have demonstrated that, in this model, skilled and empowered attorneys can successfully stack the jury with favorably-inclined jurors, but that this is equally true for the defense as for the prosecution, and further, that this is not necessarily accomplished by manipulating the racial composition of juries in predictable ways. Finally, we have verified the validity of our model empirically and found that its main prediction holds in our sample of felony trials in four large and diverse counties.

Our results are surprising in relation to previous work and to received wisdom, but less so in light of the more realistically sophisticated attorney behavior for which we allow. The attorneys in our sample end up altering the racial composition of the seated jury relative to the jury pool very little when empowered: equivalent to no more than one juror of a given race in either direction on a twelve-person jury. The implication in the context of our model is that, for at least some of the trials in our dataset, empowerment in voir dire allowed attorneys to uncover valuable information about jurors and to avoid using racial stereotypes as a crutch.

An important question is how our results might generalize to other regions not represented in our dataset. History is replete with examples of black defendants being convicted by all-white juries, and with first-hand accounts of prosecutors using peremptory challenges to remove each of the few black panelists summoned for potential jury duty only to be perfunctorily sent back home. Such examples do not invalidate our model or its predictions so much as they underline the necessity of exploring its comparative statics more carefully. In some counties, blacks may have very strong predispositions towards the defense on average, while making up a very small fraction of the population and juror roll. Other counties may be very racially diverse, but have very little variation in average predispositions across racial groups. In either of these examples, the benefit from empowerment for even the most

⁵⁰The figure in 2004 was about 4% of nearly 84,000 felony defendants, according to the Bureau of Justice Statistics.

skilled defense attorney is likely to be minimal, even if empowerment yields very detailed information about the jurors. This would especially be the case if minorities are underrepresented in jury pools and typical court procedures routinely make a large enough number of peremptory challenges available that the prosecution can easily remove any minority panelists. This situation may aptly describe many small jurisdictions, particularly in the south. It cannot be handled well within our current modeling framework. We therefore suspect that the strength of our empirical results – and perhaps the broader validity of our model – will not extend outside of large and racially diverse counties such as the ones represented in our dataset, and are leery of making any claims to the contrary without substantial further work.

Notwithstanding these concerns about generalizability, however, we do think that our present work can make important contributions to policy discussions. Many commentators have raised concerns about jury selection practices, both from the perspective of potential unfairness to the defendant and potential discrimination against minority panelists. These commentators are virtually unanimous in their calls for the abolition of peremptory strikes, which are currently employed by the American justice system to a much greater degree than anywhere else in the world.⁵¹ We submit that peremptory challenges are something of a red herring in these discussions. Specifically, our results suggest that inequality in attorney skill may be of equal importance – where this can be interpreted broadly as inequality in the resources and support available to each side for a given trial.

It would seem that collecting greater amounts of juror information should only aid in accomplishing the intended aim of voir dire – literally, to induce panelists to "speak the truth" about their biases and ability to be impartial. However, if one side lacks the skill or resources to interpret such information, this opens the door for opposing counsel to abuse this advantage by, far from seeking impartial jurors, seeking the most favorably-inclined jurors that can be slipped past the other side. Gobert et al. (2009) have suggested that large courthouses should employ analysts that both sides for a given trial can have equal access to in gaining assistance with jury selection. This seems like a worthwhile proposal to explore further, in that it could preserve the benefits of collecting a large amount of information. On the other hand, if this only aided attorneys in arguing in greater detail over semantic points rather than in actually learning about the panelists, the time and resource costs of the court would perhaps need to be given greater consideration. Again, however, our results suggest that attorneys in our dataset utilized empowerment to acquire useful information rather than just to bolster striking success in at least some cases.

We have focused on juror race for the majority of the paper, but it should be reiterated that the model only relies on the existence of any observable characteristic that is correlated with juror leanings. What constitutes observability to attorneys is not altogether clear, but virtually all courthouses collect rudimentary information on panelists' occupations and employment. Therefore, most basic sociodemographic information should be at least partially observable. In Lehmann and Smith (2013), we

 $^{^{51}}$ See, for example, Hoffman (2006).

found an especially robust relation between average jury income and acquittals, which suggests income as an obvious additional characteristic to incorporate in our analysis. More immediately, we plan to continue generalizing our model, and to explore its comparative statics theoretically and empirically.

In a sense, the present paper can be thought of as one step towards theoretically linking the empirical strategies of Anwar et al. (2012a,b) and our work in Lehmann and Smith (2013). The former study documents a correlation between the composition of *jury pools* and verdicts; and the latter documents a correlation between the composition of *seated juries* and verdicts. The present paper is about how jury selection transforms jury pools into seated juries. Can attorneys cause the distributions by race and predispositions of people on seated juries to differ from those of people in the corresponding starting jury pools? The second step would be to go from the distributions of race and predispositions on seated juries to the verdicts that those seated juries deliver, which would entail an examination of group dynamics in jury deliberations. As an extreme example, the work of Sunstein (2000) suggests that a jury with only a few members exhibiting weak pre-deliberation leanings towards conviction could lead to "group polarization" and a unanimous decision to convict under various deliberation conditions. There are indications that group decision-making effects were at play in the results discussed in Lehmann and Smith (2013), and their potential to arise is also discussed by Anwar et al. (2012b) in interpreting their results. Bowers et al. (2001) provide a valuable catalogue of types of racial interaction that occurred during deliberations in their sample of capital trials. We also plan to address this question of deliberation dynamics more carefully in future work as part of our overall project on jury selection.

The interest amongst economists in jury selection and legal findings, though for the most part very recent, should perhaps not be surprising. The legal system presents a high-stakes environment in which a number of individuals are making constrained choices, sometimes with strategic considerations. Such an environment provides a wide array of interesting questions to economists and decision scientists. Unfortunately – and perhaps unexpectedly given the volume of legal activity passing through courtrooms on a daily basis⁵² – data that would enable such questions to be satisfactorily addressed are for the most part lacking. The few econometric studies discussed in the introduction have all made use of customized self-collected datasets of one sort or another. The dataset that we have employed is unique and of great value in the richness of the information it contains, but the limited number of trials it covers remains its deficiency. The more general point is that growth in this part of the literature is currently severely constrained by data availability, not by any dearth of interesting questions remaining to be answered.

 $^{^{52}}$ Mize et al. (2007) estimate that state courts alone conduct nearly 150,000 jury trials per year, despite a steady increase in the frequency of various non-trial dispositions – itself an interesting phenomenon.

A Appendix

A.1 Alternative Measures of Attorney Empowerment

The results in the main part of the paper make use of our main indicator of attorney empowerment. which, as discussed previously, is a dummy variable for whether either high attorney participation or a supplementary juror questionnaire were allowed. Table A.1 repeats column (4) of Table 3 with several alternatives for capturing attorney empowerment. Column (4) of Table A.1 reproduces column (5) of Table 3. The first two columns consider the two main instruments potentially available to attorneys separately. Comparing with column (4), it appears that the effect of a supplementary questionnaire is biased by the omission of the indicator for high participation. The third column presents a specification fully capturing the effects of the two instruments in any combination. The large and positive effect from high participation is again present, while the effect of having a questionnaire alone is positive but insignificant. The total net effect from having both instruments together is positive but not statistically different from zero.⁵³ The specification in column (4) – representing our preferred measure of empowerment – is equivalent to that in column (3) under two parameter restrictions: the high participation effect is equal to the questionnaire effect; and the effect from having both instruments is equal to the negative of the effect of having either individually. The null hypothesis that this composite parameter restriction holds cannot be rejected against the two-sided alternative.⁵⁴ The advantage of our chosen measure of empowerment is that it can parsimoniously yet comprehensively capture the potential that attorneys had to acquire information on jurors for any given trial. Table A.1 at least provides no evidence of a statistical basis for preferring an alternative measure. On the other hand, the same broadly holds true for measuring empowerment with the indicator for high participation alone, though it is a less comprehensive measure.⁵⁵ We have repeated selected specifications replacing our chosen measure of empowerment with high participation, and have found no substantive difference in results.

Finally, column (5) of Table A.1 includes other variables related to attorney empowerment along with our baseline measure. The availability of a higher number of peremptory strikes unequivocally increases attorneys' ability to strike additional potential jurors. However, we do not observe the *potential* number of peremptory strikes available, but only the number actually exercised by each side. When attorneys are able to strike the jurors they would like to for cause, they may have no need for the total number of peremptory strikes allowed them. Therefore, a lower number of peremptory strikes actually exercised could indicate a low number available and hence a low level of empowerment on the one hand, or a high level of empowerment and unexercised peremptory challenges on the other hand. In addition, judges have a high degree of discretion over the number of peremptory strikes allowed, and often increase the number based on the severity of the charge and other aspects of the case. The inclusion of these additional variables adds little explanatory power to the regression, and provides little insight in general.

 $^{^{53}}$ The *p*-value of the two-sided test is 0.20.

 $^{^{54}}$ The Wald statistic with two degrees of freedom is 1.99, and the associated *p*-value of the test is 0.37.

⁵⁵The specification in column (3) collapses to that in column (1) under the restriction that the coefficients for 'Questionnaire' and 'both' are jointly equal to zero. The test of this null hypothesis against a two-sided alternative yields a Wald statistic with two degrees of freedom of 2.37 and a p-value of 0.31.

Dep. Var. = Juror Predisposition $(1=P \text{ to } 7=D)$								
	(1)	(2)	(3)	(4)	(5)			
High Attorney Partipation	0.420***		0.528***					
	(0.159)		(0.172)					
Questionnaire		0.031	0.364					
		(0.163)	(0.305)					
Either				0.461^{***}	0.491***			
			0 2 04	(0.156)	(0.166)			
Both			-0.584					
Denomentary Strikes (Def)			(0.357)		0.050**			
Peremptory Strikes (Del.)					-0.039°			
Peremptory Strikes (Pros.)					-0.001			
recomposity services (1105.)					(0.013)			
Venire Size					-0.002			
					(0.005)			
Voir Dire Length (hrs)					-0.001			
					(0.012)			
			- <i></i> -					
Log-Likelihood	-3446.6	-3455.4	-3442.7	-3444.2	-3311.1			
# Jurors	1883	1883	1883	1883	1812			
# Irials	256	256	256	256	244			

Table A.1: Alternative Measures of Attorney Empowerment (Ordered Logistic Regressions)

Notes: Robust standard errors, clustered at the trial level, are in parentheses. All trials with non-missing data for at least one juror are included. Each equation includes controls for attorney skill, strength of evidence, juror characteristics, county, crime type, and defendant and victim characteristics, all of which have been suppressed in the table, along with the estimated ordered logistic cut points. *p < 0.10, **p < 0.05, ***p < 0.01, based on two-tailed z-test.

A.2 Empowerment and Jury Race: Logit Specification

Dep. Var. $= 1$ if Non-White, 0 if White								
	Judge +	Jurors Ratir	ng of Skill	Judge Rating of Skill				
	(1)	(2)	(3)	(4)	(5)	(6)		
Empowerment	0.078	0.036	0.105	0.088	0.075	-0.029		
	(0.131)	(0.175)	(0.147)	(0.128)	(0.180)	(0.155)		
Empowerment \times Defense More Skilled					0.144	0.145		
					(0.248)	(0.157)		
Defense More Skilled		0.098	-0.069	0.099	0.002			
		(0.227)	(0.134)	(0.119)	(0.183)			
LA	-1.208***	-1.203***	-1.203***	-1.180***	-1.169^{***}	-1.195^{***}		
	(0.217)	(0.216)	(0.217)	(0.218)	(0.218)	(0.217)		
Maricopa	-3.095***	-3.084***	-3.097***	-3.100***	-3.079***	-3.078***		
	(0.232)	(0.234)	(0.234)	(0.232)	(0.234)	(0.233)		
DC	-1.634^{***}	-1.631^{***}	-1.623^{***}	-1.639^{***}	-1.635^{***}	-1.612^{***}		
	(0.198)	(0.198)	(0.199)	(0.202)	(0.201)	(0.201)		
Log-Likelihood	-1116.4	-1116.3	-1116.7	-1114.3	-1113.9	-1115.6		
# Jurors	1883	1883	1883	1883	1883	1883		
# Trials	256	256	256	256	256	256		

Table A.2: Prediction 2: Empowerment and Jury Race (Logit	Table A.2:	Prediction 2	2:	Empowerment	and	Jury	Race	(Logit)
---	------------	--------------	----	-------------	-----	------	------	---------

Notes: Robust standard errors, clustered at the trial level, are in parentheses. Each specification includes a constant term and controls for crime type and defendant and victim characteristics, all of which have been suppressed in the table. *p < 0.10, **p < 0.05, ***p < 0.01.

References

- Abrams, David S., Marianne Bertrand, and Sendhil Mullainathan (2011) "Do Judges Vary in their Treatment of Race?" Journal of Legal Studies, forthcoming.
- Abrams, David S. and Albert H. Yoon (2007) "The Luck of the Draw: Using Random Case Assignment to Investigate Attorney Ability," University of Chicago Law Review, Vol. 74, pp. 1145–1148.
- Alesina, Alberto F. and Eliana La Ferrara (2011) "A Test of Racial Bias in Capital Sentencing," National Bureau of Economic Research Working Paper, No. 16981.
- Altonji, Joseph G and Charles R Pierret (2001) "Employer learning and statistical discrimination," The Quarterly Journal of Economics, Vol. 116, No. 1, pp. 313–350.
- Anwar, Shamena, Patrick Bayer, and Randi Hjalmarsson (2012a) "A Fair and Impartial Jury? The Role of Age in Jury Selection and Trial Outcomes," National Bureau of Economic Research Working Paper, No. 17887.
- (2012b) "The Impact of Jury Race in Criminal Trials," The Quarterly Journal of Economics, Vol. 127, No. 2, pp. 1017–1055.
- Autor, David and David Scarborough (2008) "Does job testing harm minority workers? Evidence from retail establishments," The Quarterly Journal of Economics, Vol. 123, No. 1, pp. 219–277.
- Baldus, David C., George Woodworth, David Zuckerman, Neil Alan Weiner, and Barbara Broffitt (2001)
 "The Use of Peremptory Challenges in Capital Murder Trials: A Legal and Empirical Analysis," University of Pennsylvania Journal of Constitutional Law, Vol. 3, pp. 3–170.
- Bowers, William J., Benjamin D. Steiner, and Marla Sandys (2001) "Death Sentencing in Black and White: An Empirical Analysis of the Role of Jurors' Race and Jury Racial Composition," University of Pennsylvania Journal of Constitutional Law, Vol. 3, pp. 171–274.
- Brams, Steven J. and Morton D. Davis (1978) "Optimal Jury Selection: A Game-Theoretic Model for the Exercise of Peremptory Challenges," *Operations Research*, Vol. 26, No. 6, pp. 966–991.
- DeGroot, Morris H. and Joseph B. Kadane (1980) "Optimal Challenges for Selection," Operations Research, Vol. 28, No. 4, pp. 952–968.
- Diamond, Shari S., Destiny Peery, Francis J. Dolan, and Emily Dolan (2009) "Achieving Diversity on the Jury: Jury Size and the Peremptory Challenge," *Journal of Empirical Legal Studies*, Vol. 6, No. 3, pp. 425–449.
- Farber, Henry S and Robert Gibbons (1996) "Learning and wage dynamics," The Quarterly Journal of Economics, Vol. 111, No. 4, pp. 1007–1047.

- Feddersen, Timothy and Wolfgang Pesendorfer (1998) "Convicting the Innocent: The Inferiority of Unanimous Jury Verdicts Under Strategic Voting," American Political Science Review, Vol. 92, No. 1, pp. 23–35.
- Flanagan, Francis (2013) "Peremptory Challenges and Jury Selection." Working paper available at https://sites.google.com/site/franflanaganwfu/.
- Ford, Roger Allen (2010) "Modeling the Effects of Peremptory Challenges on Jury Selection and Jury Verdicts," *George Mason Law Review*, Vol. 17, pp. 377–381.
- Gobert, James J., Ellen Kreitzberg, and Charles H. Rose (2009) Jury Selection: The Law, Art, and Science of Selecting a Jury: Eagan, MN: West.
- Hannaford-Agor, Paula L., Valerie P. Hans, Nicole L. Mott, and G. Thomas Munsterman (2002) Are Hung Juries a Problem?: National Center for State Courts.
- (2003) "Evaluation of Hung Juries in Bronx County, New York, Los Angeles County, California, Maricopa County, Arizona, and Washington, DC, 2000–2001," National Center for State Courts User Guide. Williamsburg, VA: National Center for State Courts [producer]. Ann Arbor, MI: Interuniversity Consortium for Political and Social Research [distributor].
- Hoffman, Morris B. (2006) "Unnatural Selection," New York Times, March 6, 2006.
- Iyengar, Radha (2011) "Who's the Fairest in the Land? Analysis of Judge and Jury Death Penalty Decisions," Journal of Law and Economics, Vol. 54, No. 3, pp. 693–722.
- Kadane, Joseph B., Christopher A. Stone, and Garrick Wallstrom (1999) "The Donation Paradox for Peremptory Challenges," *Theory and Decision*, Vol. 47, pp. 139–155.
- Kadane, Joseph B. and David Kairys (1979) "Fair Numbers of Peremptory Challenges in Jury Trials," Journal of the American Statistical Association, Vol. 74, No. 368, pp. 747–753.
- King, Nancy J. (1993) "Postconviction Review of Jury Discrimination: Measuring the Effects of Juror Race on Jury Decisions," *Michigan Law Review*, Vol. 92, No. 1, pp. 63–130.
- Lee, Jean (2010) "Do Jurors Discriminate? Evidence from State Juror Selection Procedures," 5th Annual Conference on Empirical Legal Studies Paper.
- Lehmann, Jee-Yeon K. (2012) "Job Assignment and Promotion Under Statistical Discrimination: Evidence from the Early Careers of Lawyers." Working paper available at http://www.uh.edu/ ~jlehman2/.
- Lehmann, Jee-Yeon K. and Jeremy Blair Smith (2013) "A Multidimensional Examination of Jury Composition, Trial Outcomes, and Attorney Preferences," Working Paper available at http: //www.uh.edu/~jlehman2/papers/lehmann_smith_jurycomposition.pdf.

- Mize, Hon. Gregory E., Paula Hannaford-Agor, and Nicole L. Waters (2007) "The State-of-the-States Survey of Jury Improvement Efforts: A Compendium Report," National Center for State Courts, Charlottesville, VA.
- Neilson, William S. and Harold Winter (2000) "Bias and the Economics of Jury Selection," International Review of Law and Economics, Vol. 20, pp. 223–250.
- Pfeifer, J. E. (1990) "Reviewing the Empirical Evidence on Jury Racism: Findings of Discrimination or Discriminatory Findings?" Nebraska Law Review, Vol. 69, pp. 230–250.
- Roth, Arthur, Joseph B. Kadane, and Morris H. Degroot (1977) "Optimal Peremptory Challenges in Trials by Juries: A Bilateral Sequential Process," *Operations Research*, Vol. 25, No. 6, pp. 901–919.
- Shayo, Moses and Asaf Zussman (2011) "Judicial Ingroup Bias in the Shadow of Terrorism," Quarterly Journal of Economics, Vol. 126, pp. 1447–1484.
- Shinall, J. B. (2010) "Slipping Away from Justice: The Effect of Attorney Skill on Trial Outcomes," Vanderbilt Law Review, Vol. 63, pp. 267–306.
- Sommers, Samuel R. and Phoebe E. Ellsworth (2003) "How Much Do We Really Know about Race and Juries? A Review of Social Science Theory and Research," *Chicago-Kent Law Review*, Vol. 78, pp. 997–1031.
- Starr, V. Hale and Mark McCormick (2001) Jury Selection, Third Edition: Aspen Publishers Online.
- Sunstein, Cass R. (2000) "Deliberative Trouble? Why Groups Go to Extremes," The Yale Law Journal, Vol. 110, No. 1, pp. 71–119.